

## Crustal Structures from receiver functions and gravity modeling in Central Mongolia

Alexandra Guy, Christel Tiberi, Saandar Mijiddorj

## ► To cite this version:

Alexandra Guy, Christel Tiberi, Saandar Mijiddorj. Crustal Structures from receiver functions and gravity modeling in Central Mongolia. Journal of Geophysical Research: Solid Earth, 2023, 129 (1), pp.e2023JB027614. 10.1029/2023JB027614. hal-04503688

## HAL Id: hal-04503688 https://hal.science/hal-04503688

Submitted on 13 Mar 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

<mark>,</mark>



# **JGR** Solid Earth

## **RESEARCH ARTICLE**

10.1029/2023JB027614

#### **Key Points:**

- Seismic and gravity analyses confirm in 3D the relamination of a hydrous material in southern Mongolian collage
- We found no evidence of a Precambrian basement block beneath the Hangay dome
- Fragments of early Tonian plutons are imaged as an alternation of low-velocity and low-density layers with high-density material

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

A. Guy, alexandra.guy@geology.cz; alexandra.guy@gmail.com

#### Citation:

Guy, A., Tiberi, C., & Mijiddorj, S. (2024). Crustal structures from receiver functions and gravity modeling in central Mongolia. *Journal of Geophysical Research: Solid Earth*, *129*, e2023JB027614. https://doi. org/10.1029/2023JB027614

Received 6 AUG 2023 Accepted 6 DEC 2023

#### **Author Contributions:**

Conceptualization: Alexandra Guy, Christel Tiberi Data curation: Saandar Mijiddorj Formal analysis: Alexandra Guy, Christel Tiberi Funding acquisition: Alexandra Guy, Christel Tiberi Investigation: Alexandra Guy, Christel Tiberi, Saandar Mijiddorj Methodology: Alexandra Guy, Christel Tiberi Project Administration: Alexandra Guy, Christel Tiberi Resources: Alexandra Guy, Christel Tiberi, Saandar Mijiddorj

#### © 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## **Crustal Structures From Receiver Functions and Gravity Modeling in Central Mongolia**

## Alexandra Guy<sup>1</sup>, Christel Tiberi<sup>2</sup>, and Saandar Mijiddorj<sup>3</sup>

<sup>1</sup>Czech Geological Survey, Centre for Lithospheric Research, Prague, Czech Republic, <sup>2</sup>Géosciences Montpellier, CNRS-UMR5243, Université de Montpellier, Montpellier, France, <sup>3</sup>MONMAP Engineering Services, Ulaanbaatar, Mongolia

Abstract 3D forward gravity modeling combined with receiver function (RF) analysis characterizes the crustal structures of the southern part of the Mongolian collage. The seismic signals of the 48 stations of the MOBAL2003 and the IRIS-PASSCAL experiments were analyzed to get the RFs. This analysis revealed a significant difference between the crustal structures of the Hangay dome and the tectonic zones in the south. In addition, seismic stations south of the Hangay dome display significant signals related to the occurrence of a low-velocity zone at lower crustal level confirmed by the gravity anomalies. Finally, these seismic analysis inputs, the boundaries, the lithologies, and the density values from rock samples of the different tectonic zones constitute the starting points from the 3D forward gravity modeling. The resulting crustal density model indicates: (a) the likely absence of a Precambrian basement block beneath the Hangay dome, (b) an alternation of two low-velocity/low-density zones (LVLDZs) with high-density zones in the Baydrag microcontinent interpreted as fragments of early Tonian plutons, (c) the occurrence of an LVLDZ at the lower crustal level beneath the Lake zone, the Mongol-Altai Accretionary Wedge, and the Trans-Altai Zone. Therefore, the combination of the seismic RF with gravity analysis and modeling reveals new crustal structures of the Mongolian collage and enhances the occurrence and the extent of an LVLDZ at lower crustal level. These LVLDZ may demonstrate the existence of the relamination of a hydrous material in southern Mongolian collage.

**Plain Language Summary** We combined geophysical data sets to obtain a 3D model of the southern part of Mongolia. We found that the relamination of continental rocks in the lower crust may explain successive contrasted layers in this accretionary domain. We used seismic data from 48 stations to create a crustal thickness map and compared it to a 3D modeling of gravity data. We discovered significant differences between the Hangay dome's crust and the rest of the area, including the likely absence of a Precambrian basement block beneath Hangay. We also found a low-velocity and low-density zone in the lower crust beneath the southern tectonic zones, which we interpret as the relamination of a previous volcanic arc beneath the southern part of Mongolia, and question a more regional extent for this crust composition change process.

### 1. Introduction

Since plate tectonics is accepted, for modern Earth crust, geoscientists tend to clearly distinguish the oceanic crust made of basalts and gabbros, from the more complex continental crust made of metamorphosed rocks, granites, and sedimentary sequences (e.g., Hawkesworth & Kemp, 2006). Thus, at a global scale, the oceanic crust is considered dense, whereas the continental crust is lighter, due to their compositions. This also implies that the former has a faster seismic wave velocity than the latter. A three-layer crust with seismic waves and densities increasing with depth is then the typical schematic geophysical view (e.g., Christensen & Mooney, 1995; Rudnick & Fountain, 1995). However, the situation can be far more complex in the context of an accretionary orogen, as it involves the imbrication of oceanic crusts with continental crusts (Cawood et al., 2009). Crustal low-velocity and low-density zones can be locally identified and correspond to either tectonic or magmatic processes (Gilligan et al., 2015; Hauser et al., 2007; Janik et al., 2009; Kumar et al., 2022; Zorin et al., 2002). The nature and the layering of the crust together with the orientation of the interfaces are key factors to understand the interaction between the different processes involved in the geodynamics of complex accretionary orogen. Exploring the structure of crustal layers and the deep part of the orogenic lithosphere is often dedicated to the sake of seismic methods. However, the seismic coverage is often scarce and one study method often brings bias due to the limits of the technique. As they depend on the density of the rocks, gravity data serve as a proxy to define the rocks in



Supervision: Alexandra Guy, Christel Tiberi Validation: Alexandra Guy, Christel Tiberi Writing – original draft: Alexandra Guy, Christel Tiberi Writing – review & editing: Alexandra Guy, Christel Tiberi the studied volume of the crust and reduce the uncertainty of the crustal structures. Thus, density and seismic velocity models can be combined in order to identify regions with highly hydrated minerals and to distinguish the oceanic and the continental domains into an accretionary orogen, leading to a better-constrained crustal model.

The Central Asian Orogenic Belt (CAOB) formed due to accretionary followed by collisional processes and is located from the Oural to the Okhotsk Sea and from the North China to the Arctic Ocean. It is considered as a fossil mountain belt and its relics are the product of a long-lasting amalgamation of oceanic and continental crusts. The accretionary phase ranged from Neoproterozoic to Permian (Sengör et al., 1993; Wilhem et al., 2012) ending with magmatic events leading to large Late Carboniferous to Permian magmatic provinces (Jahn, 2004). The N-S-oriented collisional phase lasted from Late Permian to Triassic involving the Siberian Craton in the north and the North China and Tarim Cratons in the south (e.g., Xiao et al., 2009; Yakubchuk et al., 2005). Then, continental extension probably due to the roll-back of the Pacific plate mostly affected the eastern part of the CAOB in the Cretaceous (Charles et al., 2012; Daoudene et al., 2009, 2012; Darby et al., 2004; Meng, 2003; Ren et al., 2002). Since the Cenozoic, it has been evolving as an intracontinental orogen and the Permian-Triassic high-strain zones are reactivated as seismically active transpressive faulting and thrusting (Meltzer et al., 2019; Rizza et al., 2015; Walker et al., 2007) due to the southern India-Eurasia collision (Calais et al., 2003). Thus, different crustal types and lithological units compose the CAOB, such as the continental blocks, passive margins, oceanic domains, accretionary wedges, magmatic arcs, and back-arc basins (Kröner et al., 2010). The Proterozoic to Palaeozoic tectonic units located south of the Siberian Craton and corresponding to Mongolia, northern China, southern Russia, and eastern Kazakhstan form the so-called "Mongolian Collage" (Xiao et al., 2015), since all the representative tectonic units occur in Mongolia. Thus, Mongolia is at a pivotal position in the CAOB and our study area is located to the southern part of the Mongolian collage. Over the past two decades, extensive work has been carried out in the CAOB to characterize its geological, geochemical, and geochronological evolution at both local (a few kilometers) and regional (hundreds of kilometers) scales (e.g., Buslov et al., 2004; Janoušek et al., 2018; Lehmann et al., 2010; Xiao et al., 2015). However, only a few geophysical studies have been performed at the crustal scale (Comeau et al., 2018, 2020; Mordvinova et al., 2007; Nielsen & Thybo, 2009; Petit et al., 2002).

Recent studies based on magnetic and gravity analyses demonstrated that the mosaic of terranes defining the CAOB tectonics do not have geophysical equivalent in the southwestern part of the Mongolian Collage (Guy, Schulmann, Munschy, et al., 2014; Guy et al., 2020). Subsequently, crustal-scale models further constrained by geology and magmatic geochemistry were performed (Guy et al., 2015, 2021). These results located an old (800–500 Ma) allochthonous felsic to intermediate juvenile homogeneous material at lower crustal level beneath a younger (400–350 Ma) geophysically heterogeneous upper crust of the CAOB. Therefore, the continental felsic material formed outside of the system, before being tectonically redistributed. This feature is interpreted as a result of a trench-directed lower crustal relamination of a Precambrian to Cambrian arc beneath the Devonian oceanic crust. This process most probably took place during the late Devonian. However, the presence of relaminant in the lower crust was demonstrated only along three 2D modeling profiles located in southern and western parts of the Mongolian Collage, which brought the conclusion more at a local than global scale.

We propose combining 3D gravity modeling and analysis with seismic constraints through the prism of receiver functions (RFs) to characterize the structures of the southern Mongolian collage, part of the CAOB accretionary orogen, and assess the geodynamic processes. Seismic coverage comes down with one broad-band seismic station profile, namely MOBAL 2003 (Mordvinova et al., 2007), completed with recent large-scale temporary seismic CMSE experiments (Meltzer et al., 2019). We will particularly re-evaluate the RFs of 48 seismic stations located north of the previous geophysical and geological study following the protocol of Zhu and Kanamori (2000) and Frederiksen and Bostock (2000). Thus, the seismic inputs give crustal thickness, orientations of the seismic interfaces, the changes of velocities through these interfaces, as well as the boundaries of the tectonic zones. Then, 3D forward modeling of the Bouguer gravity anomalies constrained by geology, gravity data, density measurements, and the RF analyses will characterize the crustal structures and bring a new consistent 3D crustal architecture of the southern Mongolian Collage.

## 2. Background on the Crustal Structures of Central Mongolia

First, the main tectonic zones of the southern part of the Mongolian collage and their relevant lithologies are described. They constitute a significant source for the gravity and seismic signals. Second, we review the previous





**Figure 1.** Tectonic and geological maps (modified from Parfenov et al. (2003) and Wang et al. (2019)). (a) Global tectonic map of the Mongolian collage with the location of the 3D forward modeling area highlighted by the red box. The black curves are the state borders. The dark red area underlines the assumed extent of the Ikh-Mongol Arc remnants. (b) Geological map of Central Mongolia (modified from Parfenov et al. (2003) and Guy, Schulmann, Munschy, et al. (2014)) with the superimposed tectonic zones.

gravity and seismic studies over our study area. Their lithospheric scale was mostly dedicated to the Mesozoic and Cenozoic volcanism and the investigation of the timing and the source of the doming in the Hangay.

## 2.1. Tectonic and Geological Settings

The representative tectonic units of the CAOB located south of the Siberian craton constitute the Mongolian collage (Xiao et al., 2018), whose southern part includes Central Mongolia. It can be divided into six tectonic zones (Figure 1): (a) the Mongol-Okhotsk domain in the north is composed of Silurian to Carboniferous oceanic

sediments, which are surrounded by (b) the Mongolian Precambrian continental blocks including to the north the Tarvagatay, to the west the Zavkhan and to the south the Baydrag blocks; (c) the Bayankhongor Complex is intercalated between the southern part of the Mongol-Okhotsk domain and the Precambrian blocks, and it is composed of an ophiolitic mélanges obducted within Neoproterozoic and Cambrian accretionary wedges; (d) the Lake Zone (LZ) in the south of the Baydrag block, which is composed of Cambrian-Ordovician arc, also called Ikh-Mongol Arc, developed on Late Proterozoic accretionary complexes; (e) the Mongol-Altai Accretionary Wedge (MAAW) in the south being Cambrian-Ordovician in age; (f) and the Trans-Altai Zone (TAZ) in the southernmost part of the study area, which is an oceanic domain, composed of Devonian to Carboniferous volcano-sedimentary sequences.

The Mongol-Okhotsk oceanic domain formed during Silurian-Carboniferous and progressively closed eastward from Permian to early Cretaceous with N and S-directed, bi-vergent subduction (Bussien et al., 2011; Cogné et al., 2005; Zonenshain et al., 1990). Its western part consists of an oceanic plate stratigraphy with Siluro-Devonian pelagic to prominent Devonian-Carboniferous turbiditic deposits (Badarch et al., 2002; Kurihara et al., 2009; Tomurtogoo et al., 2005) deformed into a synclinorium and intruded by early Permian to Triassic granitoids (Filippova, 1969; Hara et al., 2013; Jahn et al., 2004; Yarmolyuk et al., 2008). Since the Cenozoic, the volcanic activity has been located at the periphery of the Hangay Dome. Some studies assumed that the Mongol-Okhotsk sedimentary cover developed on a Precambrian basement, called the Hangay microcontinent (Buchan et al., 2001; Osozawa et al., 2008; Purevjav & Roser, 2012; Zorin et al., 1993).

The Bayankhongor Complex is composed of ophiolites formed at  $569 \pm 21$  Ma (Kepezhinskas et al., 1991) and accreted at 540 Ma, intercalated in between a Silurian passive margin in the north (Buchan et al., 2001) and a Neoproterozoic to Ordovician accretionary prism in the south (Buchan et al., 2001; Jian et al., 2010). However, the tectonic context of the emplacement of the ophiolites is still under debate as Buchan et al. (2001) demonstrated that the ophiolites mark the suture between the Baydrag and the Hangay microcontinents, whereas Osozawa et al. (2008) said that the ophiolites were obducted in an accretionary complex. The south-dipping South Hangay sinistral fault zone is interpreted as a major crustal boundary between the Bayankhangor Complex and the Baydrag microcontinent (e.g., Buchan et al., 2001; Comeau et al., 2018; Walker et al., 2007).

The Baydrag microcontinent is part of the Mongolian Precambrian blocks. It contains Proterozoic-Grenvillian tonalitic granulites and granitic gneiss that are unconformably overlain by Neoproterozoic to early Cambrian metasediments, intruded by Cambrian-Ordovician metaigneous rocks, overlain by Devonian to Permian volcaniclastic rocks (Buriánek et al., 2017; Demoux, Kröner, Badarch, et al., 2009). It formed in Paleoproterozoic and recorded the formation of Rodinia, its likely break-up, the formation of Gondwana and the Palaeo-Pacific subduction initiation, with the succession of the early Tonian magmatic arc, followed by the late Tonian passive margin deposits, and the Cambrian-Ordovician continental magmatic arc developed on the Precambrian basement (Buriánek et al., 2017; Ling et al., 2021). At the western and northern periphery of our studied area stand, respectively, the Zavkhan and the Tarvagatay Precambrian microcontinents, which constitute the continuation of the Baydrag microcontinents. The Zavkhan microcontinent is also characterized by Proterozoic high-grade igneous rocks and Neoproterozoic to Carboniferous metasediments and sediments (Bold et al., 2016; Kozakov et al., 2014; Soejono et al., 2023) and as well as for the Tarvagatay (Badarch et al., 2002; Kozakov et al., 2011; Kröner et al., 2014). The Cambro-Ordovician Ikh-Mongol Arc developed at the southern continental margin of Baydrag and constitutes a part of the LZ (Janoušek et al., 2018). The relics of this arc are assumed to be spread over more than 1.800 km along the Precambrian continental blocks (Figure 1a).

The LZ is a Neoproterozoic to early Cambrian accretionary complex thrust over the Zavkhan-Baydrag basement (Buriánek et al., 2017; Štípská et al., 2010; Zonenshain & Kuzmin, 1978). It is interpreted as the Grenvillian active margin of the southern border of Zavkhan-Baydrag microcontinents (Buriánek et al., 2017). This accretionary wedge was then intruded by the Cambrian-Ordovician Ikh-Mongol arc system (Janoušek et al., 2018), then covered by post-accretion Devonian sedimentary sequences (Kröner et al., 2010) and Carboniferous to Permian volcanic rocks. The south-dipping sinistral Cenozoic Bogd fault is often assumed to divide the LZ from the MAAW (Badarch et al., 2002; Baljinnyam et al., 1993; Kurtz et al., 2018; Rizza et al., 2011; Valtr & Hanzl, 2008). However, according to the tectonics and geological analyses, the Bogd fault crosscuts typical LZ lithologies and does not constitute a major Paleozoic tectonic contact (Demoux, Kröner, Liu, & Badarch, 2009).

The MAAW is a thick Cambrian-Ordovician volcano-sedimentary accretionary complex (Badarch et al., 2002; Jiang et al., 2017; Soejono et al., 2018; Sukhbaatar et al., 2022) covered by Silurian-Devonian volcano-clastic and

carbonate sediments interpreted as a proximal passive margin deposit (Buriánek et al., 2022; Zonenshain, 1973; Zorin et al., 1993). The MAAW was then affected by Devonian-Carboniferous magmatism, high-grade metamorphism, and anatexis (Broussolle et al., 2015; Jiang et al., 2016; Kozakov et al., 2002; Kröner et al., 2010; Sukhbaatar et al., 2022). The north dipping dextral Trans-Altai Fault zone marks the boundary between the MAAW and the TAZ (Cunningham et al., 1996). This fault is the southeastern continuation of the Bulgan fault.

The TAZ is a Silurian to Carboniferous oceanic domain with Devonian-early Carboniferous arcs and backarc basins, which comprises deep marine sediments, volcanic rocks and volcano-clastic sediments (Lamb & Badarch, 2001; Nguyen et al., 2018; Ruzhentsev et al., 1985, 1992; Xiao et al., 2009; Zonenshain, 1973). The MAAW provided parts of the Devonian-Carboniferous volcano-clastic material (Guy et al., 2020; Kröner et al., 2010; Soejono et al., 2017, 2018), which lead to the interpretation that the MAAW Silurian-Devonian passive margin sequences and the TAZ oceanic crust sequences testify to the progressive formation of the Paleo-Asian Ocean (Lamb & Badarch, 2001; Zonenshain, 1973; Zorin et al., 1993).

The three major faults we briefly described above, namely, the South Hangay, the Bogd, and the Trans-Altai faults, have poor determinations in their depth continuities. However, many studies concluded that these faults are reactivated and developed along older Paleozoic weak zone structures, which probably imply lithospheric fault features (e.g., Calais et al., 2003; Cunningham, 2001; Cunningham et al., 2009; Walker et al., 2007).

#### 2.2. Crustal Architecture From Previous Seismic and Gravity Studies

The results of the MOBAL2003 seismic experiment mostly characterized the deep lithospheric structures of central and southern Mongolia. Its RF analysis demonstrated a constant crustal thickness of ~44 km in the southern part of the Hangay dome (Mordvinova et al., 2007). Moreover, the asthenospheric upwelling beneath the Hangay dome explains its long-wavelength topography (Petit et al., 2008). A 3D forward modeling of the Gravity Field and Steady-State Ocean Circulation Explorer (GOCE) satellite gravity gradient data also revealed a thick crust with a constant thickness south of the Hangay dome (Guy et al., 2017). The Hangay dome was studied by gravity-based, seismic-based, and topography-based models, which revealed strong velocity contrast within the crust and an anomalous low-density body at the Moho (i.e., in the uppermost mantle and/or lower crust) beneath the Hangay dome and its peripheral volcanism are most probably due to a mantle plume and is probably not driven by the India-Asia collision (Mordvinova et al., 2015; Petit et al., 2002; Tiberi et al., 2008).

The crustal structures and the major contacts between the different tectonic zones were determined along three N–S-oriented 2D gravity profiles (Guy et al., 2015; Zorin et al., 1993) and recently (Figure 2), the existence of a relaminant in southern Mongolia was demonstrated based on gravity and magnetic forward modeling constrained by whole-rock geochemical data (Guy et al., 2015, 2021).

### 3. Geophysical Data

#### 3.1. Seismic Data Set

The MOBAL 2003 data (MOngolian-BAikal Lithosphere) were acquired in the frame of a collaborative seismic experiment, between Russian, Mongolian, and French institutions. A temporary seismic network of 19 broad-band seismic stations was deployed from the south of the Siberian craton to the Gobi-Altai range in southern Mongolia, building a 1,000-km-long profile (Figure 1a). The Central Mongolia Seismic Experiment (CMSE) consists of 112 broad-band seismic arrays that cover 900  $\times$  600 km<sup>2</sup> in two separate deployments between 2012 and 2016 for a cumulative 21-month record. This experiment collaborates with Lehigh University, the University of Florida, and the Institute of Astronomy and Geophysics, Mongolian Academy of Sciences (Meltzer et al., 2019), and complements the MOBAL transect (Figure 3a).

The stations recorded teleseismic earthquakes and, by using traveltime tomography as well as P-to-S RF method, information on the lithospheric structures in Central Asia was obtained (Feng, 2021; Mordvinova et al., 2007; Petit et al., 2008; Tiberi et al., 2008; Zhao et al., 2021). The previous studies focused on the first-order signal and were able to characterize the main interfaces (Moho, LAB) beneath profiles or wider regions. Here, we mainly used the radial component of the RFs, while investigating the potential dip interfaces of the crust through the pattern of the transverse component.

Zorin et al. (1993) Petit et al. (2002) Guy et al. (2015)





Figure 2. Spatial coverage of the previous seismic and gravity studies superimposed on the gray-shaded topography.

Mordvinova et al. (2007, 2015)

· - Tectonic boundaries

We focused in priority on the relaminated arc, that is, the southern tip of MOBAL profile (TSET to ALTA, Figure 1) and the 44 easternmost stations of CMSE. Using those stations allows the analysis of the along-strike variations of structures in the lower crust (Figure 3a).

## 3.2. Gravity Data Set

The 2,847 ground gravity station data are provided by MonMap Company (convention between Czech Geological Survey and MonMap engineering company). The 10-25 km spaced gravity stations were acquired in the frame of Russian-Mongolian collaborative work in the 90s, in addition to the N-S oriented profile where the stations are 2 km spaced. The rest of the gravity data and specifically the densely measured profiles with 2–2.5 km spaced gravity stations were acquired by MonMap Company. Here, we present the map of the observed gravity data, leveled and corrected from the tide (Figure 3b). The gravity data need to be corrected from the topography effect to investigate the gravity anomalies at the crustal level.

## 4. Receiver Function Analysis

## 4.1. Processing of the Receiver Function

RFs are time series that show the relative response of Earth's structure beneath the seismic station. They are computed from three-component seismograms, rotated into ZRT or LQT system coordinates (Kind et al., 2012; Langston, 1977). When a P wave encounters a velocity discontinuity at depth, part of its energy will convert into an S-wave (hereafter called Ps), mainly recorded on the radial R (ZRT) or Q (LQT) component, and which





Figure 3. Central Mongolia seismic and gravity data sets. (a) Location of the seismic stations from MOBAL 2003 (Mordvinova et al., 2007) and IRIS-PASSCAL (Meltzer et al., 2019) experiments superimposed on the topography map from Earth2014 (Hirt & Rexer, 2015). (b) Location of the gravity stations and the related raw gravity anomaly map leveled and corrected from the tide effect.

polarity reflects the sign of the velocity contrast: positive (negative) if the velocity increases (decreases) with depth (Langston, 1979). If the interface is horizontal and isotropic, the Ps phase will only be present on the radial component (R or Q). In case of anisotropy or dipping interface, however, the P-wave will also produce energy on the transverse (T) component (e.g., Jones & Phinney, 1998). Anisotropy and dipping interfaces result in timing and amplitude variations which are sinusoidal in the backazimuth for radial and transverse components. Both can also yield reverse polarity, where dips exceed the incidence angle, for example, examining the transverse component and its backazimuth variation is then of prime interest to identify potential dipping or anisotropic pattern in the crustal or subcrustal layers.

Moho depth for this area has been first estimated by Mordvinova et al. (2007) through different Vs depth profiles. We computed the RF for the seven southernmost stations of the French Mongolian Baikal Lithosphere seismic experiment (MOBAL2003) combined with 44 stations from the Central Mongolia Seismic Experiment in the study area (Figure 3a). We selected teleseismic events (Mb > 6.0) with epicentral distance ranging between  $30^{\circ}$  and  $90^{\circ}$  and filtered the original data between 0.08 and 0.8 Hz. The three components were rotated into the LQT system to ensure the maximum energy on the radial (Q) component, and we obtained the Q and the T RF using an iterative time-domain deconvolution with a Gaussian filter of 3.0 (Ligorría & Ammon, 1999). We select data with more than 80% of signal recovery, which eventually leads to minimum of 22 and maximum of 109 correct RFs per station (Table 1).

#### 4.2. Determination of the Moho Depth

We proceed with the data set with the stacking method of Zhu and Kanamori (2000) to obtain the Moho depth and a *Vp/Vs* ratio estimate beneath each of the 48 sites. This inversion allows a combined estimate of crustal depth (*H*) and *Vp/Vs* ratio using a grid search on the radial component. It takes advantage of the Ps-converted wave and its multiples to reduce the inherent trade-off between *H* and *Vp/Vs* ratio. We used an a priori crustal P-wave velocity of 6.4 km.s<sup>-1</sup> similar to the value of Mordvinova et al. (2007) and close to the one in He et al. (2016). The converted phase weighting is set to 0.6, 0.3, and 0.1 for Ps, PpPs, and PpSs + PsPs, respectively (Zhu & Kanamori, 2000). We estimate the deviation of our results relative to the data quality and the a priori P-wave crustal velocity with a bootstrap algorithm technique (Tiberi et al., 2005). We first calculate standard error from the inversion of 200 random subsets of data to estimate the variability of the results within the data set for a given crustal velocity (Vp = 6.4 km.s<sup>-1</sup>, two last columns in Table 1). This helps to assess the data set coherency and to



Table 1

Results for the 48 Stations Stacking Process of the Receiver Functions

			Elevation	Events		Moho depth	Uncertainty from Vp		Uncertainty from data set	
Station	Latitude	Longitude	(km)	nb	Vp/Vs	(km)	H (km)	Vp/Vs	H (km)	Vp/Vs
HD01	46.8722	102.861603	1.714	93	1.75	43.7	±2.0	±0.01	±0.6	±0.01
HD02	46.886398	102.406898	1.633	78	1.72	45.4	±2.1	±0.01	±0.7	±0.01
HD03	46.515701	102.547203	1.878	81	1.71	47.3	±1.8	±0.01	±1.3	±0.02
HD04	46.349602	102.093597	2.21	76	1.75	46.8	±2.1	±0.01	$\pm 0.8$	±0.01
HD05	46.124599	101.593903	2.01	89	1.74	47.6	±2.1	±0.01	±1.1	±0.02
HD06	46.233299	100.747299	1.895	91	1.73	46.5	±2.1	±0.01	$\pm 0.8$	±0.02
HD07	45.9445	100.886597	1.7	87	1.7	45.5	±2.0	±0.01	$\pm 0.6$	±0.01
HD08	45.716801	100.730202	1.576	98	1.73	44.2	±2.1	$\pm 0.01$	±1.2	±0.03
UULA										
HD09	45.413898	100.5718	1.401	80	1.74	45.6	±2.0	±0.01	$\pm 0.6$	±0.02
HD10	46.416698	100.826103	2.084	93	1.76	49.1	±2.3	±0.01	$\pm 1.8$	±0.03
HD11	46.5937	100.907501	2.233	132	1.76	50.4	±2.5	±0.01	±1.9	±0.03
OVGO										
HD12	46.692699	100.9412	2.36	100	1.77	50.8	±2.1	$\pm 0.01$	±0.7	±0.01
HD13	46.799999	100.903801	2.534	84	1.76	50.7	±2.5	$\pm 0.01$	$\pm 1.0$	±0.02
HD14	46.875801	100.807701	2.516	95	1.78	49.3	±2.0	±0.01	±2.2	±0.04
HD15	47.072601	100.949097	2.186	95	1.8	45.0	±2.2	±0.01	±1.5	±0.03
TUSG										
HD16	47.189098	101.012703	2.018	74	1.73	45.4	±2.2	$\pm 0.01$	±0.7	±0.02
HD17	47.284801	101.157898	1.908	92	1.74	44.9	±2.0	$\pm 0.01$	±0.4	$\pm 0.01$
HD18	47.3825	101.290604	1.807	94	1.74	44.6	±1.9	$\pm 0.01$	±0.7	$\pm 0.01$
HD19	47.514702	101.322601	1.788	82	1.72	45.2	±2.2	±0.01	±0.8	±0.01
HD20	47.736698	101.451401	1.6	29	1.71	45.0	±1.9	$\pm 0.08$	±2.7	±0.05
HD21	47.967701	101.433296	1.776	51	1.76	44.4	±1.2	±0.02	±0.8	±0.02
HD22	47.134102	101.752502	1.839	67	1.76	43.4	±2.0	$\pm 0.01$	±0.4	$\pm 0.01$
HD23	47.509899	100.532501	2.033	90	1.72	46.6	±2.1	$\pm 0.01$	±1.0	±0.02
HD24	47.305	100.070099	2.255	80	1.76	46.7	±2.1	$\pm 0.01$	±1.2	±0.02
HD25	47.003502	99.406097	2.058	93	1.73	51.1	±2.3	±0.01	±3.4	±0.06
HD26	47.1796	98.736298	2.557	89	1.74	49.4	±2.2	±0.01	±0.8	$\pm 0.01$
HD27	47.319199	98.400703	2.547	85	1.75	50.5	±2.3	±0.01	±0.9	±0.02
HD33	47.593899	98.022003	2.601	95	1.74	51.6	±2.4	±0.01	±0.8	±0.01
HD34	47.728001	98.335602	2.568	93	1.75	49.6	±1.9	±0.01	±0.7	$\pm 0.01$
HD35	47.844501	98.339897	2.452	78	1.76	48.7	±2.4	±0.01	±1.1	±0.02
HD36	47.890999	98.448402	2.313	87	1.78	46.7	±2.1	±0.01	±0.4	±0.01
HD37	48.0159	98.626602	2.329	76	1.73	49.7	±2.0	±0.01	±0.7	±0.01
HD44	47.849201	99.427597	2.138	44	1.77	46.5	±2.1	±0.01	±1.2	±0.02
HD45	47.675201	99.139801	2.292	80	1.72	48.3	±2.5	±0.01	±0.5	±0.01
HD46	46.828499	100.059097	2.068	82	1.77	49.3	±2.2	±0.01	±1.5	±0.03
HD47	46.145901	100.144203	1.985	90	1.76	44.8	±1.9	±0.01	±0.7	±0.01
HD56	47.4025	98.650497	2.567	48	1.71	52.2	±2.1	±0.01	±0.9	±0.01
HD63	47.958401	99.744102	2.169	91	1.72	47.0	±2.0	±0.01	±1.1	±0.02

Table 1       Continued										
			Flevation	Events		Moho depth	Uncertainty from Vp		Uncertainty from data set	
Station	Latitude	Longitude	(km)	nb	Vp/Vs	(km)	H(km)	Vp/Vs	H (km)	Vp/Vs
HD65	47.7547	100.2435	2.058	97	1.74	45.3	±2.1	±0.01	±0.5	±0.01
HD66	46.8134	98.086998	2.256	86	1.82	47.7	±2.1	$\pm 0.01$	±2.2	±0.04
HD67	46.349701	99.051201	2.192	96	1.74	47.6	±2.0	$\pm 0.01$	±0.5	±0.01
HD68	44.403	102.444702	1.707	109	1.74	42.5	±2.1	$\pm 0.01$	±3.9	±0.07
HD69	44.747898	100.399597	1.609	107	1.77	45.2	±2.0	$\pm 0.01$	±0.4	±0.01
HD70	45.389599	98.029297	2.352	98	1.75	52.8	±2.3	$\pm 0.01$	$\pm 1.0$	±0.01
DALA	45.0094	100.47	1.827	22	1.75	48.0	±2.0	$\pm 0.01$	±1.4	±0.03
ALTA	44.702	100.333	1.575	24	1.71	46.6	±2.2	$\pm 0.01$	$\pm 0.8$	±0.03
TSET	47.482	101.447	1.747	42	1.75	44.5	±1.8	±0.01	$\pm 0.8$	±0.02
BUMB	46.134	100.807	1.854	22	1.73	46.3	±2.0	±0.01	±1.6	±0.03

sort out azimuthal variation or noisy data as sources of disparity. A second source of discrepancy within the inversion is tested through 200 inversions with crustal P-wave velocities randomly picked between 6.0 and 6.8 km.s<sup>-1</sup>. In Table 1 (uncertainty from Vp column), we report the dependency of the result on the ad hoc velocity model. These two bootstrap analyses show that uncertainty in our results mainly comes from crustal velocity influence (Table 1). In our case, a 5% decrease (respectively increase) of velocity will reduce (respectively increase) the crustal thickness by about 6%–8%. The Vp/Vs ratio will only change by ~1%. The data set accounts for about 2% in average on the H uncertainty, with a maximum of 9% for HD68.

In most cases, the crustal thickness is retrieved within 2 km of uncertainty, except for HD25 and HD68 station, which presents a higher standard deviation (>3 km) due to the signal azimuthal variability. For these two stations, Moho multiples are hardly detectable, probably due to interference with either lithospheric interface or intracrustal multiples.

For most of the stations, the Moho signal is clearly associated with a positive and sharp velocity gradient (at  $\sim$ 6–7 s), and its first and second multiples are identified between  $\sim$ 19 and 21 s and  $\sim$ 24–26 s, respectively (Figure 4; Figure S1 in Supporting Information S1). For some stations, in particular HD09, Ps converted wave at the Moho exhibits a weaker amplitude that may be flattened by interference with strong intracrustal conversions. From the stacking method, the Moho depth varies from 42.5 to 52.8 km (Figure 5) for an average value of 47.2 km. *Vp/Vs* ratio ranges from 1.70 to 1.82 with a mean value of 1.75. Those values are coherent with He et al. (2016) for their westernmost area (near 104° longitude).

Except for OVGO and TUSG, the relative Moho variations along the profile are similar to Mordvinova et al. (2007) study. Total crustal thickness variation deduced from the stacking method along the profile (6 km) is less than the 9 km identified by Mordvinova et al. (2007). As no error estimate is available from their study, we estimated their crustal thickness directly from their Vs velocity profiles and we cannot rigorously compare. The differences can also be due to the inversion techniques themselves, which differ. In our case, we consider a common and constant crustal P-wave velocity of  $6.4 \text{ km.s}^{-1}$ , whereas they consider a complex multi-layer crustal structure. Thus, we do not account for lateral geological heterogeneities, which can explain the discrepancy for the northern stations of the profile. The highest difference with Mordvinova et al. (2007) study is found for TUSG and OVGO stations, where an LVZ is proposed at ~30–35 km depth. However, those two stations are reprocessed and complemented in our study with HD11 and HD15 from CMSE, which show remarkable coherency in the seismic signal. The uncertainty drops to  $\pm 3$  km for these two sites (Table 1).

A depth-migrated P-wave RF profile is presented (Figure 6). The P RFs are back-projected along the ray paths using IASP91 velocity model, and stacked within  $3 \times 3 \text{ km}^2$  bins (Kind et al., 2012). We identify continuous P-to-S conversion at the Moho, and several other crustal interfaces from South to North through the different tectonic zones. The migrated profile and the normalization of its Ps amplitude reveal two main trends of the





**Figure 4.** Radial receiver functions for four stations associated with pivotal tectonic zones (a—Mongol-Altaï, b—Baydrag, and c—Hangaï). The origin time (t = 0 s) refers to the direct P-wave arrival time. The receiver functions are sorted by backazimuth, and the P to S converted phase at the Moho (Ps) is surrounded by the black rectangle. The alternance of negative and positive pics between t = 0 and the Ps arrival time indicates crustal interlayering with potential low velocity layers.

seismic signal with a boundary located in the middle of the Baydrag microcontinent. The MOO and the northern part of Baydrag crust (HD08-HD07) have a rather homogenous seismic signal, whereas the southern Baydrag, the LZ, and the MAAW display a contrasting and alternating seismic signal (HD09 and HD69; Figure 6).

The Moho clearly deepens beneath the Hangay dome (north of HD06 and South Hangay fault, >50 km), following the topographic variation and coherent with previous seismic analyses (Feng, 2021). The crust thins southwards to reach ~45 km with a Moho signal blurred and unfocused in the Baydrag zone beneath HD09 station. This apparent thickening of the crust-mantle boundary is certainly coming from a backazimuth variation effect, which is clearer for the non-smoothed version of the migrated profile (Figure 6), and which also appears in the



Figure 5. Results from Zhu and Kanamori (2000) stacking analysis of the receiver functions. (a) Moho depth (km). (b) Vp/Vs ratio.





**Figure 6.** Depth migrated profile (middle panel) and CCP profile (bottom panel) along the transect N12° centered in HD10 station. Ps amplitude has been normalized. Top panel represents the topography along the profile (thick black line), and the gray area corresponds to the envelope for topography 30 km apart from the transect. Stations (black triangles) are projected along the transect.

data (Figure 4b). In addition, HD09 displays contrasted and high amplitude intracrustal reflections, which are also retrieved beneath HD08 and ALTA in the MAAW. Those interfaces are associated with strong negative signals that we relate to negative velocity contrasts within the crust (low-velocity zones).

#### 4.3. Intracrustal Interfaces Determination

Aside from a clear Moho signal, some of the analyzed stations exhibit a distinct and coherent negative polarity signal at  $\sim 2-3$  s (Figure 4). This is particularly the case for HD09, where this intracrustal signal is stronger in amplitude than the Ps Moho conversion. These negative peaks are often associated with transverse azimuthal variations (Figure S2 in Supporting Information S1), and may reflect a decreasing velocity with depth associated with a dip and/or anisotropic material. This behavior is particularly clear for the southern part of the migrated profile (MAAW, Lake, and Baydrag zones, Figure 6), with an enhanced amplitude below HD09 (Baydrag).

We investigate the characteristics of these interfaces and potential LVZ through a series of inversions following the method of Frederiksen et al. (2003). The inversion is stochastic and offers the possibility to retrieve both dipping layers (thickness, *V*p, *V*s, dip, and strike) and their anisotropy behavior (%*V*p, %*V*s, trend, and plunge). However, the increase of model space parameters will result in a less constrained solution. We decided to look only for layer thickness, *V*p, *V*s, dip, and strike within the crust. Even if methods allow to investigate the anisotropic behavior of crustal part (Feng, 2021; Feng & Diaz, 2023; Shen et al., 2013), the anisotropy is too complex in our multilayer model to be constrained. Besides, we are looking for intracrustal information we can combine with gravity signal, and anisotropy will be of no help in our case.

We apply the inversion on HD09 station, which presents the most coherent radial component and which overlays potential relaminated material (Guy et al., 2015). Among all our tests, the best results are obtained for



ip

Table 2           Preferred Models From the Inversion of HD09 Receiver Functions							
Layer number	Thickness (km)	<i>V</i> p (km.s <sup>-1</sup> )	<i>Vs</i> (km.s <sup>-1</sup> )	Strike	Dip		
1	0.9	6.0	3.3	-	-		
2	4	5.1	3.1	208	14		
3	8.6	6.2	3.7	104	3		
4	0.2	5.0	2.0	45	10		

Mantle		8.1	4.25	107	3
6	11.9	5.9	3.4	135	2
5	9	6.6	3.7	129	1
•	, <b>.</b>		0.0		10

Note. Low-velocity layers (LVZs) are indicated in gray, and inverted parameters are in bold type

a six-custral layers structure overlaying a semi-infinite mantle. The mantle velocities are fixed ( $Vp = 8.1 \text{ km.s}^{-1}$ ,  $Vs = 4.25 \text{ k.s}^{-1}$ ), as well as the two first layers to reduce the number of unknowns. Those values are deduced from direct modeling to fit the two first seconds of the radial RF analysis (Table 2, Figure 4). The inversions show very little change for the four inverted layer thicknesses (Figure S2 in Supporting Information S1). The P-wave and S-wave velocities converge to coherent values, except for layer 6 which exhibits some deviation in the results (less constrained). Low-velocity layers are present in all models, but with disparity in strike, dip, or velocity contrasts. Common features are a superficial and thin LVZ (1-5 km deep), which fits the first seconds of the RF signal, a second LVZ 10-12 km thick at about 12 km depth and a third LVZ just above the Moho, 10-12 km thick. The inversion brings only little constraint on the dip and strike of these layers. Even if the transverse component displays a coherent pattern (Figure S2 in Supporting Information S1), the number of layers prevents from having a simple well-identified solution due to the large degree of freedom and to the complex interferences between the conversions. From our preferred models, all dips are less than 15° and more likely South or Southwest.

## 5. Gravity Analysis

The gravity anomalies are used to investigate the density contrasts and decipher the crustal structures, particularly in lower crustal parts. Before correlating the gravity signal with the different tectonic units and performing the density modeling, the measured data has to be corrected from the effect of topography to get the complete Bouguer gravity anomalies. Then, the geometry and density distribution of the crust can be investigated by building a 3D forward gravity model constrained by the geological and RF data.

#### 5.1. Gravity Processing

The complete Bouguer gravity anomaly grid is obtained from the free-air anomalies by computing the terrain correction of each gravity station (Data Set S1). The terrain correction was performed using a mean crustal density of 2,670 kg/m<sup>3</sup> by the GEEC Matlab-based code (Cattin et al., 2015; Saraswati et al., 2019), which takes into account the sphericity of the Earth and the global effect of the Earth's gravity field. Three different digital elevation models (DEMs) were used to test the sensitivity of the results according to the resolutions of the DEM and the gravity data: (a) Earth 2014 global topography model (Hirt & Rexer, 2015) of 1 min arc resolution (~2 km); (b) Shuttle Radar Topography Mission 30 (SRTM) of 30 s arc resolution (~1 km); and (c) SRTM1 (Farr et al., 2007) of 1 s arc resolution ( $\sim$ 30 m).

The data sets and the complete Bouguer gravity results were extracted along the N-S profile, where most of the seismic stations are located, to examine the similarities and the discrepancies between the topography models, the results of the complete Bouguer anomalies, and the different seismic Moho depth (Figure 7). The three topography models are equivalent from the southern part of the Mongol-Okhotsk domain to the MAAW but display significant discrepancies from 0 to 165 m in the rest of the Mongol-Okhotsk domain, which corresponds to the high-frequency topography relief of the Hangay Dome. Thus, the Earth2014 model has a lower amplitude in this area than the SRTM30 and the SRTM1. The raw gravity data are logically anti-correlated with the topography. The differences in the frequency of the topography between the topography models are in return impacting the terrain correction results, as the complete Bouguer anomalies obtained with the Earth2014 model have lower amplitude than those obtained with SRTM30 and SRTM1. The comparison of the resulting complete Bouguer anomalies also shows that the use of SRTM1 does not necessarily bring strong differences in the terrain correction taking into account the resolution of the gravity data. Thus, for future computations, using SRTM30 may be enough and for sure a gain of time. In the last section, we display and discuss the differences between the Moho depths obtained by GOCE satellite gravity gradient modeling, the previous seismic RF analyses and our seismic results (Figure 7). The discrepancies observed between the seismic Moho and the gravity Moho depth from GOCE gravity gradient data are quite significant. In addition, the different seismic studies show rather poor agreement, particularly for the seismic stations OVGO and TUSG, which are the seismic stations located in the





Figure 7. Comparison of the different data sets, their resulting processing, and previous studies. The N–S profile extracted in all data set. along with Mordvinova et al. (2007). Profiles comparing the topography of three different topography models, the raw gravity data, the resulting complete Bouguer anomalies, and the seismic Moho depth from our study and from previous studies.

Hangay Dome. These differences are probably due to the methods and assumptions used to get the Moho depth, as explained in the seismic section.

The resulting complete Bouguer gravity anomaly map with a grid spacing of 10 km ranges from -310 to -185 mGal (Figure 8). The Trans-Altai fault displays a mild gravity gradient, whereas the South Hangay fault correlates with a strong gravity gradient. The Bogd fault also displays gravity contrasts, which are not really linear but in the shape of "boudins." Finally, the gravity gradient located at the northeastern part of the map corresponds to





Figure 8. Complete Bouguer anomaly map with the lithological contacts and the tectonic zones, the main faults, and the location of seismic stations superimposed.

a fault, which presents significant seismic events from 0 to 15 km depth (Meltzer et al., 2019) and identified as an Early Mesozoic rift system (Yarmolyuk & Kuzmin, 2012).

#### 5.2. Constraints and Implementation of the Model

The gravity modeling of the central Mongolian crust is constrained by the depths of the Moho obtained by the RF analyses of this study, complemented by the topography of the Moho obtained by the 3D forward modeling of the GOCE gravity gradients (Guy et al., 2017). In addition, the modeling is constrained by geological data such as the surface boundaries of the different tectonic zones and their characteristic lithological composition (e.g., metamorphic, magmatic, or sedimentary rocks) (Figure 1), the dip of the contacts determined by previous studies (Badarch et al., 2002; Buchan et al., 2001; Cunningham et al., 1996; Sukhbaatar et al., 2022; Walker et al., 2007; Zorin et al., 1993). Thus, these benchmarks bring constraints on the shape and the depth of the main crustal density units. The RF inversion brought only poor constraints on the dip of the contact, and the signal's frequency content does not permit high detail level in the structure. We thus only took first-order information from these inversions: average depth location of low-velocity zones and interfaces.

The model also benefits from density values measured on 343 rock samples collected in the different tectonic zones (Figure 9; Table S1). The density values range from 2,430 to 3,312 kg/m<sup>3</sup> and the mean rock density of 2,739 kg/m<sup>3</sup> is obtained with a standard deviation of 145 kg/m<sup>3</sup>. However, the Gaussian of the frequency diagram gives a pick at 2,680 kg/m<sup>3</sup>. The highest density average distribution correlates with the location of the Precambrian metaigneous garnet-bearing amphibolites in northern Baydrag and the lowest density average distribution corresponds with the Devonian-Carboniferous turbiditic basin in the Mongol-Okhotsk domain. The intermediate density values from 2,680 to 2,750 kg/m<sup>3</sup> are found in the MAAW and the TAZ containing volcano-sedimentary sequences related to their oceanic environments. However, the constraints on the density values remain poor for a given lithology, surely metamorphosed in the deeper part of the crust and thus, the seismic velocities and the





Figure 9. Density sample locations and average distributions according to the distinct tectonic zones used for the modeling. Values in kg/m<sup>3</sup>.

geological model are integrated to assess the most suitable density. Table 3 summarizes the density values that were assigned in our 3D forward gravity model.

The 3D forward gravity modeling was performed with the IGMAS+ software package (Anikiev et al., 2020; Gotze & Lahmeyer, 1988; Schmidt et al., 2011). The 3D block model is built by 32 north-south oriented 2D parallel sections with spacing of 15 km, which contains the density units linked by polyhedrons with triangular surfaces connecting the vertices between the successive sections. The model covers an area of  $\sim$ 480 km  $\times$  400 km and reaches 70 km in depth. First, as the topography of the Moho is constrained by the seismic RF results and the previous GOCE gravity gradient model (Guy et al., 2017), a simple homogeneous crust-mantle model with standard density values was generated in order to observe the contribution of the variations of the Moho within the long-wavelengths of the gravity signal (Figure 10a). Thus, we can see that the Moho density contrast does not explain the entire gravity signal. Second, the crust was divided into six tectonic zones, each split into the conventional upper, middle, and lower crusts (Christensen & Mooney, 1995), with densities increasing with depth (Rudnick & Fountain, 1995). During this process, the dips of the main contacts between the distinct tectonic zones were maintained and the average density values were attributed according to the characteristic surface lithologies observed for each tectonic zone. At this step, the gravity signal is still not correctly modeled (Figure 10b). Finally, the model gained complexity by including the geological models (Buchan et al., 2001; Guy et al., 2015, 2021; Zorin, 1999; Zorin et al., 1993) and the interfaces obtained by RF analyses. As the target is to model the structures of the crust, we did not take into account the possible variation of densities in the upper mantle and we assumed a laterally homogeneous upper mantle of 3,200 kg/m<sup>3</sup>.

#### 5.3. 3D Forward Gravity Modeling Results

The residual map of the final 3D model (Figure 11a) shows that the highest amplitudes of the residuals are at the western edge of the model. This border effect can be explained by the changes of lithologies in the western periphery of the model from Mongol-Okhotsk oceanic domain to the Zavkhan Precambrian microcontinent, whereas the geological units continue to the eastern periphery of the model (Figure 11b). The average misfits between the measured and the calculated anomalies have a correlation coefficient of 09.975 and a standard deviation of 2.35 mGal.

The N–S trending profile extracted from the 3D model is 530 km long (Figure 11c). Its location has been defined as it perpendicularly crosses the main tectonic zones and it runs along the area where most of the seismic stations from the MOBAL2003 and IRIS-PASSCAL experiments are located. The density structure profile starts at the southern margin of the Tarvagatay microcontinent, crosses the Mongol-Okhotsk oceanic domain, the Bayankhangor Complex, the Baydrag microcontinent, the LZ, the MAAW, and ends in the northern part of the TAZ. The complete Bouguer gravity anomalies range from  $\sim$ -215 mGal in the south to -290 mGal in the center of the Mongol-Okhotsk domain. The long wavelength gravity signal along the profile can be divided into



Tectonic zones	Lithologies	Densities [kg/m <sup>3</sup> ]
Tavargatay block		
	Devonian-Carboniferous sediments	2650
	Neoproterozoic sediments	2680
	Archean-Neoproterzoic basement	2850
Mongol-Okhotsk domain		
	Devonian-Carboniferous Turbidites	2650
	Silurian basement	2740
Bayankhangor Complex		
	Ophiolites	2730
	Neoproterozoic metamorphic complex	2700
	Cambro-Ordovician metamorphic complex	2700
Baydrag microcontinent		
	Proterozoic metasediments	2700
	Low Velocity Zone 1	2650
	Low Velocity Zone 2	2750
	Proterozoic crystalline basement 1	2800
	Proterozoic crystalline basement 2	2850
	Granitoid	2650
Lake Zone		
	Relics of Ikh-Mongol arc	2770
	Proterozoic basement	2710
	Ordovician metasediments	2650
MAAW		
	Ordovician metasediments	2760
	Silurian metasediments	2740
	Devonian volcano-sediments	2670
	Carboniferous sediments	2650
	Permian sediments	2620
Trans-Altai Zone		
	Siluro-Devonian volcano-sediments	2750
	Carboniferous sediments	2640
Permian-Triassic granitoids		2640
Felsic lower crust		2730
Mantle		3200

two main trends: a high amplitude and low-frequency gravity over the MOO and the northern Baydrag, and a low amplitude and medium frequency gravity anomalies over the southern Baydrag, the LZ, the MAAW, and the TAZ (Figure 11c). The Moho depth, constrained by the RF analysis and the GOCE gravity gradient modeling (Guy et al., 2017), varies from  $\sim$ -45 to 50 km depth along the profile and can be considered as rather flat in its southern part as it varies from -46.6 to -48 km. The upper crust of the southern Paleozoic tectonic zones, namely the LZ, the MAAW, and the TAZ, is modeled, taking into account the compositions of the different lithologies, as laterally variable nearly horizontal density bodies, which reproduce the large-scale folding and thrusting of the accretionary system cut off by steep contacts between each tectonic zone (Guy, Schulmann, Clauer, et al., 2014; Lehmann et al., 2010; Sukhbaatar et al., 2022). These features correspond to the short-wavelength anomalies observed in the southern part of the profile. The nearly constant long-wavelength anomaly is interpreted as the





**Figure 10.** Modeling constraints and implementation of the model. The profile presented here is extracted from the 3D block along the N–S profile, where most of the seismic stations are located. The resulting residual maps of the corresponding 3D forward gravity modeling are shown for each implementation. Densities are in  $kg/m^3$ . (a) Crust-mantle gravity model with standard density values. The resulting Moho depth from seismic receiver function and in between the Moho depth from GOCE satellite gravity gradient 3D modeling are depicted. (b) Gravity model with conventional three-layer crust and density values increasing with depth. The tectonic boundaries are displayed.

lack of density variation in the lower crust, since the Moho depth does not significantly oscillate. Thus, the lower crust underneath the LZ, the MAAW, and the TAZ has a homogeneous gravity signature and is characterized by a felsic density value of 2,730 kg/m<sup>3</sup> (Guy et al., 2015, 2021). The Bogd fault located in the LZ can be identified by a small gradient in the gravity signal and is modeled accordingly. The gravity analysis and modeling point toward a slight crustal discontinuity between the Baydrag microcontinent and the LZ as exemplified by the short wavelength gravity low at the contact, which can however correspond to the presence of Ordovician metasediments in the Proterozoic basement of the LZ or to the presence of the Ikh-Mongol arc relics east and west of the profile (Figure 11c). Geological constraints concerning the Baydrag microcontinent are rather poor so far. We can





Figure 11. Results of the 3D gravity model. (a) Illustration of the final 3D block model. (b) Residual anomaly map of the model. (c) N–S profile extracted along the seismic stations. The characteristics of the density bodies are given in the legend. Densities are in kg/m<sup>3</sup>.

just assert that there is a significant difference between the southern, seismically heterogeneous, and the northern part, seismically homogeneous (Figure 6). Thus, the Baydrag microcontinent is modeled as a multi-layered basement crust in its southern part, where two low-density layers alternate with higher-density layers according to the resulting seismic interfaces, and a rather homogeneous to two-layer crust with densities of 2,700 and 2,850 kg/m<sup>3</sup> in its northern part, with an undetermined steep boundary in the middle of the microcontinent. The Bayankhongor Complex is 20 km wide in the N–S direction and the gravity data have 10 km spaced measurements in this area. Thus, it has a poor resolution and could not have been modeled indeed. Nevertheless, according to geological studies (Buchan et al., 2001; Osozawa et al., 2008) and to maintain a tectonic coherency, the Bayankhangor Complex is modeled as three vertical density units reproducing the ophiolitic mélange intercalated in volcano-sedimentary to sedimentary clastic sequences. Finally, the Mongol-Okhotsk domain is modeled as a two-layer sedimentary crust, which represents the Silurian sedimentary sequences overlain by the Devonian-Carboniferous turbiditic basins together folded into a synclinorium pattern, later intruded by Permian-Triassic granitoids. The Cenozoic sedimentary basins in central Mongolia are supposed to be approximately 0–2 km thick and short-wavelength anomalies do not spatially correlate with them at the location and scale of our 3D modeling area. Therefore, they were not included in the model.

21699356, 2024, 1, Downloaded from https:

online library.wiley.com/doi/10.1029/2023JB027614 by Cochrane France

Wiley Online Library on [13/03/2024]. See

### 6. Discussion

The following section examines the seismic and gravity patterns of the crust in Central Mongolia by first discussing the similarities and differences of our crustal model with the electromagnetic profiles acquired in the same area and the significance of these geophysical anomalies. Then, we analyze the distribution of the seismic interfaces from the RFs and the geometry of the density structures from the forward modeling, related to surface geology of the different tectonic zones. These correlations enhance new crustal structures, such as middle and lower crust low-velocity/low-density zones (LVLDZs) in the Precambrian block and the Paleozoic domains, and challenge some boundaries between the tectonic zones. Thus, we demonstrate the complementarity of seismic and gravity methods to decipher the crustal structures in complex, reworked, and long-lasting tectonic settings.

#### 6.1. Comparison of Seismic and Gravity Results With Electromagnetic Profile

To evaluate our crustal model based on seismic and gravity data sets, we compare it with recent N-S-oriented electromagnetic profiles (Comeau et al., 2018, 2020), whose data were acquired nearly at the same location where most of the seismic stations from the MOBAL2003 and IRIS-PASSCAL experiments are distributed (Figure 12). In general, the electrical resistivity models of the crust reveal a high resistive upper crust (0–20 km) with several conductive features, that are interpreted as fault or suture zones, and a low resistive lower crust, interpreted as a presence of fluids and a weakened lower crust.

Specifically, the Mongol-Okhotsk upper crust displays a variable resistivity (10–2,000  $\Omega$ m) in its first 0.5 km (C1<sub>(2018)</sub>, Figure 12b) probably due to the occurrence of porous sediments. Below, the upper crust presents rather highly resistive zones (~10,000  $\Omega$ m) interspersed with vertical channels (C4<sub>(2018)</sub>) of lower resistivity (400–1,500  $\Omega$ m), which correlate with location of volcanism and present-day hydrothermal activity. The MOO lower crust has a low resistivity zone between 30 and 50 km (20–80  $\Omega$ m), indicating the presence of fluids and a weakened lower crust  $(C2_{(2018)})$ . These two electrical layers of the MOO, each rather homogeneous, concur with the homogeneous seismic signal of the MOO crust (Figure 6) and the two density layers of the 3D forward model (Figure 12a). The electrical resistivity profile reveals a major crustal boundary in the shape of a low resistivity zone (20-40  $\Omega$ m) concerning the South Hangay fault system and the ophiolitic belt (F1(2018)) interpreted as the marker of hydrothermal alteration along fossil fluid pathways due to the location of a suture zone (Comeau et al., 2021). Ophiolites have to provide distinct gravity and seismic signals due to their compositions (Christensen, 1978). The Bayankhangor ophiolitic complex is a NW-SE belt of cca. 300 km long and cca. 20 km wide presumably obducted within two accretionary systems, so that a strong seismic and gravity signal should be expected. However, the Bayankhongor complex does not exhibit strong seismic interfaces (HD6, HD7, and BUMB; Figure 6). Moreover, only the southern part of the complex correlates with a gravity high centered on the ophiolitic mélange; otherwise, no distinct gravity high can be distinguished along its entire length (Figure 8). This lack of seismic interface and gravity highs may be related to the spacing of gravity and seismic stations or because the ophiolites may be only superficial and not deeply rooted (Figure 12a). Similarly, the South Hangay fault is not associated with strong seismic and gravity signal anomalies, which may indicate a shallow depth of the contact and a low-density contrast between the juxtaposed lithologies, respectively. Comeau et al. (2018, 2020) interpreted the shallow conductive features (<5 km) in the southern Baydrag microcontinent as a significant mineralization and past hydrothermal fluid alteration leading to the Tsagaan Tsahir Uul gold deposit. Their electrical model also shows a low resistivity zone starting from 20 km in the northern and 10 km in the southern parts of the Baydrag microcontinent with a vertical heterogeneous resistivity up to 20 km depth in between (G1(2020)). Overall, the northern part of the Baydrag microcontinent appears more resistant than the southern part. This trend is similar to our seismic and gravity results showing that the Baydrag microcontinent is geophysically divided into two parts: the northern part does not display strong seismic interfaces, whereas the southern part presents an alternation of low-velocity with high-velocity zones respectively corresponding to low-density with high-density layers. The Bogd fault in the LZ correlates with a low resistivity anomaly  $(30-100 \Omega m)$  dipping to the south (F2<sub>(2018)</sub>). However, the apparent dip angle of the fault given by the electrical model is  $\sim 30^{\circ}$  (Comeau et al., 2018), whereas the gravity modeling results in a subvertical dipping fault (Figure 12a). Moreover, no seismic contrast enables the detection of the Bogd fault, although it was identified as a deeply rooted (up to 20 km depth) active fault (Kurushin et al., 1998; Rizza et al., 2011). This poor resolution of the fault may result from the seismic coverage and the orientation of the fault, which might be too vertical to generate an RF signal. Finally, the distinct zones of low resistivity material (50–100  $\Omega$ m) located at 30–45 km depth beneath the LZ and the MAAW  $(C1_{(2020)})$  and the low resistivity zone (~100  $\Omega$ m) at 15–30 km depth in the TAZ  $(C2_{(2020)})$  are in agreement with the





**Figure 12.** Crustal cross-section through the southern Mongolian collage. (a) Profile extracted from the 3D forward gravity model at the location of the EM profile, with seismic receiver function cones of key stations superimposed on the gravity model; (b) Corresponding electrical resistivity model modified from Comeau et al. (2018, 2020).  $C1_{(2018)}$ : variable resistivity (10–2,000  $\Omega$ m);  $C4_{(2018)}$ : (400–1,500  $\Omega$ m) vertically elongated;  $C2_{(2018)}$ : Low resistivity zone (20–80  $\Omega$ m) below the Hangay Dome between 30 and 50 km;  $F1_{(2018)}$ : Low resistivity zone (20–40  $\Omega$ m);  $G1_{(2020)}$ : Shallow (<5 km) conductive feature;  $F2_{(2018)}$ : Low resistivity zone (30–100  $\Omega$ m) between 30 and 50 km;  $C2_{(2020)}$ : Low resistivity zone (~100  $\Omega$ m) between 15 and 30 km.

identification of the LVLDZ in our seismic and gravity models (Brownish density body of 2,730 kg/m<sup>3</sup>, Figure 12a), although the RF results give a shallower low-velocity zone starting from 20 km up to the Moho in the southern part of the profile. Therefore, the electrical resistivity model and our model based on seismic and gravity data converge regarding the boundaries of the different tectonic zones and their upper and lower crustal structures.

#### 6.2. Crustal Structures of the Mongol-Okhotsk Oceanic Domain

The Mongol-Okhotsk oceanic domain is the Paleozoic tectonic zone, also corresponding to the Cenozoic Hangay Dome, which is an intracontinental high-elevation low-relief landform in Central Mongolia. The Hangay Dome is surrounded by Mesozoic and Cenozoic intraplate volcanism attributed to the asthenospheric upwelling (Ancuta et al., 2018; Hunt et al., 2012; Sheldrick et al., 2018). Recent geochemical and geophysical studies were under-taken to decipher the mechanisms of such a dome formation and intraplate volcanism in intracontinental tectonic settings (e.g., Ancuta et al., 2018; Comeau et al., 2021; Tiberi et al., 2008). The thickness of the southern Mongolian crust is rather constant with the seismic and gravity Moho located at ~45 km depth (Figures 6 and 11), which is an average continental crustal thickness considered above the estimated standard of around 39.5 km (Christensen & Mooney, 1995). This steady thickness of the crust in this part of the Mongolian collage was also demonstrated by previous seismic and gravity studies (Guy et al., 2017; Mordvinova et al., 2007; Petit et al., 2002). Then, the thickness of the crust increases in the southern part of the MOO beneath the Hangay

Dome, whereas the asthenospheric upwelling reveals the thinning of the lithosphere leading to the Cenozoic magmatism (Guy et al., 2017; Petit et al., 2002, 2008). However, the flexure of the Moho is not enough to explain the high amplitude gravity low over the MOO (Figure 10a).

Interpretations of geochemical and gravity data led some authors assuming that the Devonian-Carboniferous sedimentary basin of the MOO developed on a Precambrian basement, called Hangay microcontinent (Cunningham, 2001; Osozawa et al., 2008; Purevjav & Roser, 2012; Zorin et al., 1993). Taking into account the homogeneity of the seismic signal and the strong gravity low, these geophysical characteristics in the MOO can be explained by the combination of the Devonian-Carboniferous sedimentary layers on the Silurian meta-sediments and their magmatic reworking in Permian-Triassic with the intrusion of mantle and crustal source granitoids (Yarmolyuk et al., 2016). Although at the scale of the 3D model, no real close correlation between the location of the Permian-Triassic granitoids in the MOO and the short wavelength gravity low is generally observed, we integrated these granitoids to the model to avoid any possible bias. However, the involvement of a hypothetical Precambrian basement will not help to model the gravity low and this sedimentary layer structure is in agreement with the absence of a distinct seismic interface within the MOO, which could have been interpreted as the interface between the Paleozoic sedimentary basin and a Precambrian continental fragment. Therefore, our model may reveal the absence of a Precambrian basement in the MOO (Figure 13), previously considered beneath the Devonian to Carboniferous Hangay dome and questioned by Erdenesaihan et al. (2013), Ruppen et al. (2014), and Tsukada et al. (2013).

#### 6.3. Significance of the Low-Velocity and Low-Density Zones in the Baydrag Microcontinent

As previously mentioned, contrasting seismic signals are observed between the northern and the southern parts of the Precambrian Baydrag microcontinent (Figure 6). Its northern seismic structure does not show strong interfaces, which implies either a relative homogeneity of the lithologies or highly verticalized structures due to a high convergence rate. Hence, this was modeled as a homogeneous density body with a thin denser lower crust. Its southern seismic structure is characterized by sub-horizontal interfaces showing the alternance of low-velocity and high-velocity zones, which were modeled as a tectonic layering with the alternance of low-density and high-density bodies accordingly (Figure 11).

Geological investigations leading to a geodynamic reconstruction of the Baydrag microcontinent are rather scarce, particularly in its southern part due to the occurrence of the Cenozoic sedimentary basin, which conceals the Proterozoic and Paleozoic tectonic units. Thus, the tectonic interpretation of the Baydrag southern margin mostly comes from an assumed continuation of the tectonic trends from the Zavkhan southern margin (Soejono et al., 2023), as it belongs to the Precambrian microcontinent belts located in the center of the Mongolian collage. Nevertheless, studies converge with the probability of contrasting geodynamics affecting the northern and southern Baydrag microcontinent. The northern part of the Baydrag microcontinent is affected by an early Tonian (890–855 Ma) extension of the continental crust revealed by HT–LP metamorphism and interpreted as the consequence of the subduction roll-back of the Mirovoi Oceanic plate beneath the Rodinia (Štípská et al., 2023). This event is followed by the late Tonian thickening of a back-arc or arc domain due to the subduction of the Mirovoi Ocean beneath Rodinia, followed by a Cambro–Ordovician continental magmatic arc related to the subduction of the Paleo-Pacific Ocean beneath Gondwana (Buriánek et al., 2017; Soejono et al., 2023). This Cambro–Ordovician magmatic event is not observed in the northern part. In addition, the Tonian magmatism related to an arc was not described in the northern part of Baydrag microcontinent and will probably not be distinguished at crustal scale.

Therefore, based on the geological studies, the difference of crustal structures between the northern and southern parts of Baydrag revealed by seismic and gravity data analyses may be explained by the distinct tectonic events affecting the Baydrag microcontinent. In the southern part, the LVLDZs may constitute the remnants of the batholiths resulting from the early Tonian or the Cambro–Ordovician continental magmatic arc (Figure 13) or may result from late Tonian anatexis processes due to the extension during the break-up of Rodinia that would have affected southern Baydrag. However, this late Tonian extension related to the break-up of Rodinia was identified in all the other Precambrian microcontinents in the Mongolian collage except in Baydrag (Soejono et al., 2023). Thus, we rather assume the hypothesis of remnants of batholiths, which can partly trigger the alternation of high velocity-high density layers with low-velocity/low-density layers at the crustal scale, as it was demonstrated in the Archean Karelia and Kola provinces, where the acoustic transparent regions and low seismic reflection zone were interpreted as fragments of batholiths (Kostyuchenko et al., 2006; Mints et al., 2020).







#### 6.4. Relaminated Material Under the Mongol Altai Accretionary Wedge and the Trans-Altai Zone

It is accepted that the present continental crust is formed by magmatic processes at volcanic arcs above subduction zones (e.g., Hacker et al., 2015; Hawkesworth & Kemp, 2006). Volcanic arcs have a geochemical composition very close to the continental crust, at least in its upper part, since the deep crust of the arcs (from ~20 km below the surface) has a very different geochemical composition from the continental crust at equivalent depth. The denser part of the arc can undergo delamination, which is a loss of part of the denser lower lithosphere that sinks into the mantle and is recycled (e.g., Bird, 1979; Magni et al., 2013). However, this process is not sufficient to transform a lower arc crust into a continental crust. Delamination does occur, but for it to be the only mechanism responsible for crustal differentiation, a complex process of crustal thickening and repeated metamorphic events would be required. Thus, a new process of crustal reworking was proposed (Hacker et al., 2011). During subduction followed by continental collision, a part of the crustal lower-plate material becomes buoyant and can be redistributed within the crustal upper-plate inducing its compositional and structural transformations. The process of such transformation of the continental crust is called relamination (Hacker et al., 2011; Kelemen & Behn, 2016) and the resulting material of redistributed meta-sedimentary and meta-igneous rocks is referred to as the relaminant. Recent petrological data (Hacker et al., 2011, 2015), geochemical data (Guy et al., 2015; Kelemen & Behn, 2016; Yarmolyuk et al., 2012) and numerical modeling (Maierova et al., 2018) converge to this process of relamination.

The seismic interfaces observed in the LZ and the MAAW crusts show an alternation of increase and decrease of velocities and can be interpreted as a succession of high-velocity and low-velocity zones together with a low Vp/Vs ratio (Figures 4 and 6). The geological cross-sections proposed in these tectonic zones provide evidence of the succession of folding and thrusting of the different lithologies stemming from the magmatic and tectonic events, and sedimentary deposits affecting the LZ Cambrian passive margin and the Mongol-Altai Ordovician Accretionary Wedge (Buriánek et al., 2017; Sukhbaatar et al., 2022; Štípská et al., 2010). These folds and thrusts were roughly reproduced in the gravity model in the shapes of folded sedimentary layers in the MAAW and thrusts of metasediments on the Proterozoic basement in the LZ, which enables it to fit the observed with the calculated gravity signal (Figure 11). The geologically constrained upper parts of the LZ, MAAW, and TAZ allow us concluding that the weak amplitude of the gravity signal (~25 mGal) over these zones indicates a lack of deep crustal discontinuities (Guy, Schulmann, Munschy, et al., 2014, 2015), which is modeled as a homogeneous felsic lower crust (Guy et al., 2015, 2021). In the lower crustal part of the LZ and MAAW, the LVLDZ is interpreted as the occurrence of an allochthonous hydrous or felsic material (Figure 13), which explains the low Vp/Vs ratio (1.71 below ALTA). Our 3D geophysical model also provides important constraints for the possible extent of the relaminant at the scale of southern Mongolia, and consequently, may indicate a huge amount of material being relaminated.

## 7. Conclusions

The combination of RF with gravity data analyses and modeling, constrained by geological data, not only refines the topography of the Moho in central Mongolia, but also provides substantial insights into the crustal structures of central Mongolia and its tectonic evolution.

- The significant difference of the crustal seismic and gravity anomalies between the Mongol Okhotsk Oceanic domain and the southern tectonic zones, namely the Baydrag microcontinent, the LZ, the MAAW, and the TAZ, involves different crustal structures and testifies to the contrasting geodynamics.
- The seismic homogeneity of the Mongol Okhotsk Oceanic crust and the 3D gravity model demonstrate the absence of Precambrian microcontinent nuclei beneath the Hangay dome.
- The contrasting seismic signatures between the northern and southern parts of the Baydrag microcontinent confirm the distinct geodynamic evolution between its northern and southern margins. Moreover, the LVLDZs within the Baydrag Precambrian block middle crust can be interpreted as the remnants of the Grenvillean or Cambro-Ordovician batholiths.
- The LVLDZ at the lower crustal level beneath the LZ, the MAAW, and the TAZ reveal the occurrence of a felsic material, and, thus the existence of an allochthonous crust of continental affinities beneath an oceanic crust. This is interpreted as the result of a relamination, which is probably an underestimated tectonic process of continental crust formation.

## **Data Availability Statement**

Seismic data (Meltzer et al., 2019) used in this study are open-access and obtained from https://www.iris.edu/ hq/. The density data used in the 3D model are available as supplementary material. Satellite gravity data (Guy et al., 2017) are available from web services at https://earth.esa.int/eogateway/missions/goce. Land gravity data supporting this research are available at the MonMap engineering company and were obtained upon a formal convention between Czech Geological Survey and MonMap engineering company. The Complete Bouguer anomaly grid used in the modeling is available in supplementary material. The grid was calculated using the GEEC software (Saraswati et al., 2019; https://github.com/RodolpheCattin/GEEC). The 3D forward gravity modeling was performed using the IGMAS+ software (Anikiev et al., 2020; https://igmas.git-pages.gfz-potsdam.de/igmaspages/). The other data can be obtained in the tables and the references.



This study was funded by the PHC-Barrande project No. 42674XJ and the Grant Agency of the Czech Republic (GAČR EXPRO Grant GX19-27682X). This work was supported by CNES. It is based on GOCE mission. MonMap company provided the ground gravity data. The authors thank Pavel Hanžl and Luc De Hoÿm de Marien for their support in the sample collection for the density measurements. Denis Anikiev and Sabine Schmidt are also acknowledged for their efficient support regarding the IGMAS+ software. The authors are grateful to Karel Schulmann, Igor Soejono, and Pavla Štípská for fruitful discussions. The authors would like to thank Michael Bostock and the associated editor for editor handling, and Wenjiao Xiao and two anonymous reviewers for their constructive comments.

### References

- Ancuta, L. D., Zeitler, P. K., Idleman, B. D., & Jordan, B. T. (2018). Whole-rock 40Ar/39Ar geochronology, geochemistry, and stratigraphy of intraplate Cenozoic volcanic rocks, central Mongolia. GSA Bulletin, 130(7–8), 1397–1408. https://doi.org/10.1130/b31788.1
- Anikiev, D., Hans-Jürgen, G., Meeßen, C., Plonka, C., Scheck-Wenderoth, M., & Schmidt, S. (2020). IGMAS+: Interactive Gravity and Magnetic Application System. V. 1.3, edited, GFZ Data Services, Postdam [Software]. GFZ. https://doi.org/10.5880/GFZ.4.5.igmas.v.1.3
- Badarch, G., Cunningham, D. W., & Windley, B. F. (2002). A new terrane subdivision for Mongolia: Implications for the Phanerozoic crustal growth of Central Asia. *Journal of Asian Earth Sciences*, 21(1), 87–110. https://doi.org/10.1016/s1367-9120(02)00017-2
- Baljinnyam, I., Bayasgalan, A., Borisov, B. A., Cisternas, A., Dem'yanovich; M. G., Ganbaatar, L., et al. (1993). Ruptures of major earthquakes and active deformation in Mongolia and its surroundings. Geological Society of America. https://doi.org/10.1130/MEM181
- Bird, P. (1979). Continental delamination and the Colorado Plateau. Journal of Geophysical Research: Solid Earth, 84(B13), 7561–7571. https:// doi.org/10.1029/jb084ib13p07561
- Bold, U., Crowley, J. L., Smith, E. F., Sambuu, O., & Macdonald, F. A. (2016). Neoproterozoic to early Paleozoic tectonic evolution of the Zavkhan terrane of Mongolia: Implications for continental growth in the Central Asian orogenic belt. *Lithosphere*, 8(6), 729–750. https://doi. org/10.1130/1549.1
- Broussolle, A., Štípská, P., Lehmann, J., Schulmann, K., Hacker, B. R., Holder, R., et al. (2015). P–T–t–D record of crustal-scale horizontal flow and magma assisted doming in the SW Mongolian Altai. *Journal of Metamorphic Geology*, 33, 359–383. https://doi.org/10.1111/jmg.12124 Buchan, C., Cunningham, D., Windley, B. F., & Tomurhuu, D. (2001). Structural and lithological characteristics of the Bayankhongor Ophiolite
- Zone, Central Mongolia. Journal of the Geological Society, 158(3), 445–460. https://doi.org/10.1144/jgs.158.3.445 Buriánek, D., Schulmann, K., Hrdličková, K., Hanžl, P., Janoušek, V., Gerdes, A., & Lexa, O. (2017). Geochemical and geochronological
- constraints on distinct Early-Neoproterozoic and Cambrian accretionary events along southern margin of the Baydrag Continent in western Mongolia. *Gondwana Research*, 47, 200–227. https://doi.org/10.1016/j.gr.2016.09.008
- Buriánek, D., Soejono, I., Schulmann, K., Janoušek, V., Hanžl, P., Čáp, P., et al. (2022). Subduction-controlled temporal and spatial variations in early Palaeozoic sedimentary and volcanic record of the Mongol-Altai Domain. *Journal of Asian Earth Sciences*, 230, 105182. https://doi.org/10.1016/j.jseaes.2022.105182
- Buslov, M. M., Fujiwara, Y., Iwata, K., & Semakov, N. N. (2004). Late Paleozoic-Early Mesozoic geodynamics of Central Asia. Gondwana Research, 7(3), 791–808. https://doi.org/10.1016/s1342-937x(05)71064-9
- Bussien, D., Gombojav, N., Winkler, W., & von Quadt, A. (2011). The Mongol–Okhotsk Belt in Mongolia—An appraisal of the geodynamic development by the study of sandstone provenance and detrital zircons. *Tectonophysics*, 510(1), 132–150. https://doi.org/10.1016/j. tecto.2011.06.024
- Calais, E., Vergnolle, M., Sankov, V., Lukhnev, A., Miroshnitchenko, A., Amarjargal, S., & Deverchere, J. (2003). GPS measurements of crustal deformation in the Baikal-Mongolia area (1994-2002): Implications for current kinematics of Asia. *Journal of Geophysical Research*, *108*(B10), 2501. https://doi.org/10.1029/2002jb002373
- Cattin, R., Mazzotti, S., & Baratin, L.-M. (2015). GravProcess: An easy-to-use MATLAB software to process campaign gravity data and evaluate the associated uncertainties. *Computers & Geosciences*, 81, 20–27. https://doi.org/10.1016/j.cageo.2015.04.005
- Cawood, P. A., Kröner, A., Collins, W. J., Kusky, T. M., Mooney, W. D., & Windley, B. F. (2009). Accretionary orogens through Earth history. In P. A. Cawood, & A. Kröner (Eds.), *Earth accretionary systems in space and time* (pp. 1–36). Geological Society Special Publication.
- Charles, N., Gumiaux, C., Augier, R., Chen, Y., Faure, M., Lin, W., & Zhu, R. (2012). Metamorphic Core Complex dynamics and structural development: Field evidences from the Liaodong Peninsula (China, East Asia). *Tectonophysics*, 560–561, 22–50. https://doi.org/10.1016/j. tecto.2012.06.019
- Christensen, N. I. (1978). Ophiolites, seismic velocities and oceanic crustal structure. Tectonophysics, 47(1), 131-157. https://doi.org/10.1016/0040-1951(78)90155-5
- Christensen, N. I., & Mooney, W. D. (1995). Seismic velocity structure and composition of the continental crust: A global view. Journal of Geophysical Research, 100(B6), 9761–9788. https://doi.org/10.1029/95jb00259
- Cogné, J. P., Kravchinsky, V. A., Halim, N., & Hankard, F. (2005). Late Jurassic-Early Cretaceous closure of the Mongol-Okhotsk Ocean demonstrated by new Mesozoic palaeomagnetic results from the Trans-Bail<sup>^</sup>kal area (SE Siberia). *Geophysical Journal International*, 163(2), 813–832. https://doi.org/10.1111/j.1365-246x.2005.02782.x
- Comeau, M. J., Becken, M., Käufl, J. S., Grayver, A. V., Kuvshinov, A. V., Tserendug, S., et al. (2020). Evidence for terrane boundaries and suture zones across Southern Mongolia detected with a 2-dimensional magnetotelluric transect. *Earth, Planets and Space*, 72(1), 5. https:// doi.org/10.1186/s40623-020-1131-6
- Comeau, M. J., Becken, M., Kuvshinov, A. V., & Demberel, S. (2021). Crustal architecture of a metallogenic belt and ophiolite belt: Implications for mineral genesis and emplacement from 3-D electrical resistivity models (Bayankhongor area, Mongolia). *Earth, Planets and Space*, 73(1), 82. https://doi.org/10.1186/s40623-021-01400-9
- Comeau, M. J., Käufl, J. S., Becken, M., Kuvshinov, A., Grayver, A. V., Kamm, J., et al. (2018). Evidence for fluid and melt generation in response to an asthenospheric upwelling beneath the Hangai Dome, Mongolia. *Earth and Planetary Science Letters*, 487, 201–209. https://doi.org/10.1016/j.epsl.2018.02.007
- Cunningham, W. D. (2001). Cenozoic normal faulting and regional doming in the southern Hangay region, Central Mongolia: Implications for the origin of the Baikal rift province. *Tectonophysics*, 331(4), 389–411. https://doi.org/10.1016/S0040-1951(00)00228-6
- Cunningham, D., Davies, S., & McLean, D. (2009). Exhumation of a Cretaceous rift complex within a Late Cenozoic restraining bend, southern Mongolia: Implications for the crustal evolution of the Gobi Altai region. *Journal of the Geological Society*, *166*(2), 321–333. https://doi. org/10.1144/0016-76492008-082
- Cunningham, W. D., Windley, B. F., Dorjnamjaa, D., Badamgarov, G., & Saandar, M. (1996). A structural transect across the Mongolian Western Altai: Active transpressional mountain building in central Asia. *Tectonics*, 15(1), 142–156. https://doi.org/10.1029/95tc02354
- Daoudene, Y., Gapais, D., Ledru, P., Cocherie, A., Hocquet, S., & Donskaya, T. V. (2009). The Ereendavaa Range (north-eastern Mongolia): An additional argument for Mesozoic extension throughout eastern Asia. *International Journal of Earth Sciences*, 98(6), 1381–1393. https://doi. org/10.1007/s00531-008-0412-2
- Daoudene, Y., Gapais, D., Ruffet, G., Gloaguen, E., Cocherie, A., & Ledru, P. (2012). Syn-thinning pluton emplacement during Mesozoic extension in eastern Mongolia. *Tectonics*, 31(3), TC3001. https://doi.org/10.1029/2011tc002926
- Darby, B. J., Davis, G. A., Zhang, X., Wu, F. Y., Wilde, S. A., & Yang, J. H. (2004). The newly discovered Waziyu metamorphic core complex, YiwuliJ Shan, western Liaoning Province, Northwest China. *Earth Science Frontiers*, 11(3), 145–155.



- Demoux, A., Kröner, A., Badarch, G., Jian, P., Tomurhuu, D., & Wingate, M. T. D. (2009). Zircon ages from the Baydrag block and the Bayankhongor Ophiolite Zone: Time constraints on Late Neoproterozoic to Cambrian subduction- and accretion-related magmatism in Central Mongolia. *The Journal of Geology*, 117(4), 377–397. https://doi.org/10.1086/598947
- Demoux, A., Kröner, A., Liu, D., & Badarch, G. (2009). Precambrian crystalline basement in southern Mongolia as revealed by SHRIMP zircon dating. International Journal of Earth Sciences, 98(6), 1365–1380. https://doi.org/10.1007/s00531-008-0321-4
- Erdenesaihan, G., Ishiwatari, A., Orolmaa, D., Arai, S., & Tamura, A. (2013). Middle Paleozoic greenstones of the Hangay region, central Mongolia: Remnants of an accreted oceanic plateau and forearc magmatism. *Journal of Mineralogical and Petrological Sciences*, 108(6), 303–325. https://doi.org/10.2465/jmps.130409
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., et al. (2007). The Shuttle Radar Topography Mission. Reviews of Geophysics, 45(2). https://doi.org/10.1029/2005rg000183
- Feng, L. (2021). High-resolution crustal and uppermost mantle structure beneath Central Mongolia from Rayleigh waves and receiver functions. Journal of Geophysical Research: Solid Earth, 126(4), e2020JB021161. https://doi.org/10.1029/2020jb021161
- Feng, L., & Diaz, J. (2023). A high-resolution shear velocity model of the crust and uppermost mantle beneath westernmost Mediterranean including radial anisotropy. *Journal of Geophysical Research: Solid Earth*, 128(9), e2023JB026868. https://doi.org/10.1029/2023jb026868
- Filippova, I. B. (1969). The Khangay synclinorium: Main features of structure and evolution. Geotektonika, 5(5), 76–78.
- Frederiksen, A. W., & Bostock, M. G. (2000). Modelling teleseismic waves in dipping anisotropic structures. *Geophysical Journal International*, 141(2), 401–412. https://doi.org/10.1046/j.1365-246x.2000.00090.x
- Frederiksen, A. W., Folsom, H., & Zandt, G. (2003). Neighbourhood inversion of teleseismic Ps conversions for anisotropy and layer dip. *Geophysical Journal International*, 155(1), 200–212. https://doi.org/10.1046/j.1365-246x.2003.02043.x
- Gilligan, A., Priestley, K. F., Roecker, S. W., Levin, V., & Rai, S. S. (2015). The crustal structure of the western Himalayas and Tibet. Journal of Geophysical Research: Solid Earth, 120(5), 3946–3964. https://doi.org/10.1002/2015jb011891
- Gotze, H. J., & Lahmeyer, B. (1988). Application of three-dimensional interactive modeling in gravity and magnetics. *Geophysics*, 53(8), 1096–1108. https://doi.org/10.1190/1.1442546
- Guy, A., Holzrichter, N., & Ebbing, J. (2017). Moho depth model for the Central Asian Orogenic Belt from satellite gravity gradients [Dataset]. Journal of Geophysical Research: Solid Earth, 122(9), 7388–7407. https://doi.org/10.1002/2017JB014120
- Guy, A., Schulmann, K., Clauer, N., Hasalová, P., Seltmann, R., Armstrong, R., et al. (2014). Late Paleozoic–Mesozoic tectonic evolution of the Trans-Altai and South Gobi Zones in southern Mongolia based on structural and geochronological data. *Gondwana Research*, 25(1), 309–337. https://doi.org/10.1016/j.gr.2013.03.014
- Guy, A., Schulmann, K., Janoušek, V., Štípská, P., Armstrong, R., Belousova, E., et al. (2015). Geophysical and geochemical nature of relaminated arc-derived lower crust underneath oceanic domain in southern Mongolia. *Tectonics*, 34(5), 1030–1053. https://doi.org/10.1002/2015tc003845
- Guy, A., Schulmann, K., Munschy, M., Miehe, J.-M., Edel, J.-B., Lexa, O., & Fairhead, D. (2014). Geophysical constraints for terrane boundaries in southern Mongolia. *Journal of Geophysical Research: Solid Earth*, 119(10), 7966–7991. https://doi.org/10.1002/2014ib011026
- Guy, A., Schulmann, K., Soejono, I., Holzrichter, N., Lexa, O., & Munschy, M. (2021). Structures and geodynamics of the Mongolian tract of the Central Asian Orogenic Belt constrained by potential field analyses. Gondwana Research, 92, 26–53, https://doi.org/10.1016/j.gr.2020.11.016
- Guy, A., Schulmann, K., Soejono, I., & Xiao, W. (2020). Revision of the Chinese Altai-East Junggar terrane accretion model based on geophysical and geological constraints. *Tectonics*, 39(4), e2019TC006026. https://doi.org/10.1029/2019tc006026
- Hacker, B. R., Kelemen, P. B., & Behn, M. D. (2011). Differentiation of the continental crust by relamination. Earth and Planetary Science Letters, 307(3–4), 501–516. https://doi.org/10.1016/j.epsl.2011.05.024
- Hacker, B. R., Kelemen, P. B., & Behn, M. D. (2015). Continental lower crust. Annual Review of Earth and Planetary Sciences, 43(1), 167–205. https://doi.org/10.1146/annurev-earth-050212-124117
- Hara, H., Kurihara, T., Tsukada, K., Kon, Y., Uchino, T., Suzuki, T., et al. (2013). Provenance and origins of a Late Paleozoic accretionary complex within the Khangai–Khentei belt in the Central Asian Orogenic Belt, central Mongolia. *Journal of Asian Earth Sciences*, 75, 141–157. https://doi.org/10.1016/j.jseaes.2013.07.019
- Hauser, F., Raileanu, V., Fielitz, W., Dinu, C., Landes, M., Bala, A., & Prodehl, C. (2007). Seismic crustal structure between the Transylvanian Basin and the Black Sea, Romania. *Tectonophysics*, 430(1), 1–25. https://doi.org/10.1016/j.tecto.2006.10.005
- Hawkesworth, C. J., & Kemp, A. I. S. (2006). Evolution of the continental crust. *Nature*, 443(7113), 811–817. https://doi.org/10.1038/nature05191
  He, J., Wu, Q., Sandvol, E., Ni, J., Gallegos, A., Gao, M., et al. (2016). The crustal structure of south central Mongolia using receiver functions. *Tectonics*, 35(6), 1392–1403. https://doi.org/10.1002/2015tc004027
- Hirt, C., & Rexer, M. (2015). Earth2014: 1 arc-min shape, topography, bedrock and ice-sheet models—Available as gridded data and degree-10,800 spherical harmonics. *International Journal of Applied Earth Observation and Geoinformation*, 39, 103–112. https://doi.org/10.1016/j.jag.2015.03.001
- Hunt, A. C., Parkinson, I. J., Harris, N. B. W., Barry, T. L., Rogers, N. W., & Yondon, M. (2012). Cenozoic volcanism on the Hangai Dome, Central Mongolia: Geochemical evidence for changing melt sources and implications for mechanisms of melting. *Journal of Petrology*, 53(9), 1913–1942. https://doi.org/10.1093/petrology/egs038
- Jahn, B. M. (2004). The Central Asian Orogenic Belt and growth of the continental crust in the Phanerozoic. In J. Malpas, A. Fletcher, J. R. Ali, & J. C. Aitchison (Eds.), Aspects of the tectonic evolution of China (Vol. 226, pp. 73–100). Geological Society London Special Publications.
- Jahn, B. M., Capdevila, R., Liu, D., Vernon, A., & Badarch, G. (2004). Sources of Phanerozoic granitoids in the transect Bayanhongor-Ulaan Baatar, Mongolia: Geochemical and Nd isotopic evidence, and implications for Phanerozoic crustal growth. *Journal of Asian Earth Sciences*, 23(5), 629–653. https://doi.org/10.1016/s1367-9120(03)00125-1
- Janik, T., Kozlovskaya, E., Heikkinen, P., Yliniemi, J., & Silvennoinen, H. (2009). Evidence for preservation of crustal root beneath the Proterozoic Lapland-Kola orogen (northern Fennoscandian shield) derived from P and S wave velocity models of POLAR and HUKKA wide-angle reflection and refraction profiles and FIRE4 reflection transect. *Journal of Geophysical Research: Solid Earth*, 114(6). https://doi. org/10.1029/2008jb005689
- Janoušek, V., Jiang, Y., Buriánek, D., Schulmann, K., Hanžl, P., Soejono, I., et al. (2018). Cambrian–Ordovician magmatism of the Ikh-Mongol Arc System exemplified by the Khantaishir Magmatic Complex (Lake Zone, south–central Mongolia). Gondwana Research, 54, 122–149. https://doi.org/10.1016/j.gr.2017.10.003
- Jiang, Y. D., Schulmann, K., Kröner, A., Sun, M., Lexa, O., Janoušek, V., et al. (2017). Neoproterozoic-Early Paleozoic Peri-Pacific accretionary evolution of the Mongolian Collage System: Insights from geochemical and U-Pb zircon data from the Ordovician sedimentary wedge in the Mongolian Altai. *Tectonics*, 36(11), 2305–2331. https://doi.org/10.1002/2017tc004533
- Jiang, Y. D., Schulmann, K., Sun, M., Štípská, P., Guy, A., Janoušek, V., et al. (2016). Anatexis of accretionary wedge, Pacific-type magmatism, and formation of vertically stratified continental crust in the Altai Orogenic Belt. *Tectonics*, 35(12), 3095–3118. https://doi. org/10.1002/2016tc004271

- Jian, P., Kröner, A., Windley, B. F., Shi, Y., Zhang, F., Miao, L., et al. (2010). Zircon ages of the Bayankhongor ophiolite mélange and associated rocks: Time constraints on Neoproterozoic to Cambrian accretionary and collisional orogenesis in Central Mongolia. *Precambrian Research*, 177(1), 162–180. https://doi.org/10.1016/j.precamres.2009.11.009
- Jones, C. H., & Phinney, R. A. (1998). Seismic structure of the lithosphere from teleseismic converted arrivals observed at small arrays in the southern Sierra Nevada and vicinity, California. Journal of Geophysical Research, 103(B5), 10065–10090. https://doi.org/10.1029/97JB03540
- Kelemen, P., & Behn, M. (2016). Formation of lower continental crust by relamination of buoyant arc lavas and plutons. *Nature Geosciences*, 9(3), 197–205. https://doi.org/10.1038/ngeo2662
- Kepezhinskas, P. K., Kepezhinskas, K. B., & Pukhtel, I. S. (1991). Lower Paleozoic oceanic crust in Mongolian Caledonides: SM-ND isotope and trace element data. *Geophysical Research Letters*, 18(7), 1301–1304. https://doi.org/10.1029/91gl01643
- Kind, R., Yuan, X., & Kumar, P. (2012). Seismic receiver functions and the lithosphere-asthenosphere boundary. *Tectonophysics*, 536–537, 25–43. https://doi.org/10.1016/j.tecto.2012.03.005
- Kostyuchenko, S., Sapozhnikov, R., Egorkin, A., Gee, D. G., Berzin, R., & Solodilov, L. (2006). Crustal structure and tectonic model of northeastern Baltica, based on deep seismic and potential field data. *Geological Society*, *32*(1), 521–539. https://doi.org/10.1144/gsl.mem.2006.032.01.32
- Kozakov, I. K., Bibikova, E. V., Azimov, P. Y., & Kirnozova, T. I. (2002). Hercynian Granulites of Mongolian and Gobian Altai: Geodynamic setting and formation conditions. *Doklady Earth Sciences*, 386(7), 781–785.
- Kozakov, I. K., Kovach, V. P., Bibikova, E. V., Kirnozova, T. I., Lykhin, D. A., Plotkina, Y. V., et al. (2014). Late Riphean episode in the formation of crystalline rock complexes in the Dzabkhan microcontinent: Geological, geochronologic, and Nd isotopic-geochemical data. *Petrology*, 22(5), 480–506. https://doi.org/10.1134/s086959111405004x
- Kozakov, I. K., Kozlovsky, A. M., Yarmolyuk, V. V., Kovach, V. P., Bibikova, E. V., Kirnozova, T. I., et al. (2011). Crystalline complexes of the Tarbagatai block of the Early Caledonian superterrane of Central Asia. *Petrology*, 19(4), 426–444. https://doi.org/10.1134/s0869591111040047
- Kröner, A., Kovach, V., Belousova, E., Hegner, E., Armstrong, R., Dolgopolova, A., et al. (2014). Reassessment of continental growth during the accretionary history of the Central Asian Orogenic Belt. *Gondwana Research*, 25(1), 103–125. https://doi.org/10.1016/j.gr.2012.12.023
- Kröner, A., Lehmann, J., Schulmann, K., Demoux, A., Lexa, O., Tomurhuu, D., et al. (2010). Lithostratigraphic and geochronological constraints on the evolution of the Central Asian Orogenic Belt in SW Mongolia: Early Paleozoic rifting followed by late Paleozoic accretion. American Journal of Science, 310(7), 523–574. https://doi.org/10.2475/07.2010.01
- Kumar, V., Rai, S. S., Hawkins, R., & Bodin, T. (2022). Seismic imaging of crust beneath the western Tibet-Pamir and western himalaya using ambient noise and earthquake data. Journal of Geophysical Research: Solid Earth, 127(6), e2021JB022574. https://doi.org/10.1029/2021jb022574
- Kurihara, T., Tsukada, K., Otoh, S., Kashiwagi, K., Chuluun, M., Byambadash, D., et al. (2009). Upper Silurian and Devonian pelagic deep-water radiolarian chert from the Khangai-Khentei belt of Central Mongolia: Evidence for Middle Paleozoic subduction-accretion activity in the Central Asian Orogenic Belt. *Journal of Asian Earth Sciences*, 34(2), 209–225. https://doi.org/10.1016/j.jseaes.2008.04.010
- Kurtz, R., Klinger, Y., Ferry, M., & Ritz, J. F. (2018). Horizontal surface-slip distribution through several seismic cycles: The Eastern Bogd fault, Gobi-Altai, Mongolia. *Tectonophysics*, 734–735, 167–182. https://doi.org/10.1016/j.tecto.2018.03.011
- Kurushin, R. A., Bayasgalan, A., Ölziybat, M., Enhtuvshin, B., Molnar, P., Bayarsayhan, C., et al. (1998). The surface rupture of the 1957 Gobi-Altay, Mongolia, earthquake. Geological Society of America.
- Lamb, M. A., & Badarch, G. (2001). Paleozoic sedimentary basins and volcanic arc systems of southern Mongolia: New geochemical and petrographic constraints. In M. S. Hendrix, & G. A. Davis (Eds.), Paleozoic and Mesozoic tectonic evolution of central and eastern Asia: From continental assembly to intracontinental deformation (pp. 117–149). Memoir of the Geological Society of America.
- Langston, C. A. (1977). Crustal and upper mantle structure from teleseismic P and S waves. Bulletin of the Seismological Society of America, 67(3), 713–724. https://doi.org/10.1785/bssa0670030713
- Langston, C. A. (1979). Structure under Mount Rainier, Washington, inferred from teleseismic body waves. Journal of Geophysical Research: Solid Earth, 84(B9), 4749–4762. https://doi.org/10.1029/jb084ib09p04749
- Lehmann, J., Schulmann, K., Lexa, O., Corsini, M., Kröner, A., Štípská, P., et al. (2010). Structural constraints on the evolution of the Central Asian Orogenic Belt in SW Mongolia. American Journal of Science, 310(7), 575–628. https://doi.org/10.2475/07.2010.02
- Ligorría, J. P., & Ammon, C. J. (1999). Iterative deconvolution and receiver-function estimation. Bulletin of the Seismological Society of America, 89(5), 1395–1400. https://doi.org/10.1785/bssa0890051395
- Ling, J., Li, P., Yuan, C., Sun, M., Zhang, Y., Narantsetseg, T., et al. (2021). Ordovician to Devonian granitic plutons in the Hangay Range, Central Mongolia: Petrogenesis and insights into the Paleozoic tectonic evolution of the westernmost Mongol-Okhotsk Orogen. *Lithos*, 404–405, 106463. https://doi.org/10.1016/j.lithos.2021.106463
- Magni, V., Faccenna, C., van Hunen, J., & Funiciello, F. (2013). Delamination vs. break-off: The fate of continental collision. *Geophysical Research Letters*, 40(2), 285–289. https://doi.org/10.1002/grl.50090
- Maierová, P., Schulmann, K., & Gerya, T. (2018). Relamination styles in collisional orogens. *Tectonics*, 37(1), 224–250. https://doi. org/10.1002/2017tc004677
- Meltzer, A., Stachnik, J. C., Sodnomsambuu, D., Munkhuu, U., Tsagaan, B., Dashdondog, M., & Russo, R. (2019). The Central Mongolia seismic experiment: Multiple applications of temporary broadband seismic arrays [Dataset]. Seismological Research Letters, 90(3), 1364–1376. https://doi.org/10.1785/0220180360
- Meng, Q.-R. (2003). What drove late Mesozoic extension of the northern China–Mongolia tract? *Tectonophysics*, 369(3–4), 155–174. https://doi.org/10.1016/s0040-1951(03)00195-1
- Mints, M. V., Glaznev, V. N., Muravina, O. M., & Sokolova, E. Y. (2020). 3D model of Svecofennian Accretionary Orogen and Karelia Craton based on geology, reflection seismics, magnetotellurics and density modelling: Geodynamic speculations. *Geoscience Frontiers*, 11(3), 999–1023. https://doi.org/10.1016/j.gsf.2019.10.003
- Mordvinova, V. V., Deschamps, A., Dugarmaa, T., Deverchére, J., Ulziibat, M., Sankov, V. A., et al. (2007). Velocity structure of the lithosphere on the 2003 Mongolian-Baikal transect from SV waves, Izvestiya. *Physics of the Solid Earth*, 43(2), 119–129. https://doi.org/10.1134/ s1069351307020036
- Mordvinova, V. V., Treussov, A. V., & Turutanov, E. K. (2015). Nature of the mantle plume under Hangai (Mongolia) based on seismic and gravimetric data. *Doklady Earth Sciences*, 460(1), 92–95. https://doi.org/10.1134/s1028334x15010201
- Nguyen, H., Hanžl, P., Janoušek, V., Schulmann, K., Ulrich, M., Jiang, Y., et al. (2018). Geochemistry and geochronology of Mississippian volcanic rocks from SW Mongolia: Implications for terrane subdivision and magmatic arc activity in the Trans-Altai Zone. *Journal of Asian Earth Sciences*, 164, 322–343. https://doi.org/10.1016/j.jseaes.2018.06.029
- Nielsen, C., & Thybo, H. (2009). No Moho uplift below the Baikal Rift Zone: Evidence from a seismic refraction profile across southern Lake Baikal. Journal of Geophysical Research: Solid Earth, 114(B8), B08306. https://doi.org/10.1029/2008jb005828
- Osozawa, S., Tsolmon, G., Majigsuren, U., Sereenen, J., Niitsuma, S., Iwata, N., et al. (2008). Structural evolution of the Bayanhongor region, west-central Mongolia. *Journal of Asian Earth Sciences*, 33(5), 337–352. https://doi.org/10.1016/j.jseaes.2008.01.003



- Parfenov, L. M., Khanchuk, A. I., Badarch, G., Miller, R. J., Naumova, V. V., Nokleberg, W. J., et al. (2003). Preliminary Northeast Asia geodynamics map. Report 03-205. U.S. Geological Survey Open-File.
- Petit, C., Déverchère, J., Calais, E., San'kov, V., & Fairhead, D. (2002). Deep structure and mechanical behavior of the lithosphere in the Hangai-Hövsgöl region, Mongolia: New constraints from gravity modeling. *Earth and Planetary Science Letters*, 197(3–4), 133–149. https:// doi.org/10.1016/s0012-821x(02)00470-3
- Petit, C., Tiberi, C., Deschamps, A., & Déverchère, J. (2008). Teleseismic traveltimes, topography and the lithospheric structure across central Mongolia. Geophysical Research Letters, 35(11–16). https://doi.org/10.1029/2008gl033993
- Purevjav, N., & Roser, B. (2012). Geochemistry of Devonian–Carboniferous clastic sediments of the Tsetserleg terrane, Hangay Basin, Central Mongolia: Provenance, source weathering, and tectonic setting. *Island Arc*, 21(4), 270–287. https://doi.org/10.1111/j.1440-1738.2012.00821.x
- Ren, J., Tamaki, K., Li, S., & Junxia, Z. (2002). Late Mesozoic and Cenozoic rifting and its dynamic setting in Eastern China and adjacent areas. *Tectonophysics*, 344(3–4), 175–205. https://doi.org/10.1016/s0040-1951(01)00271-2
- Rizza, M., Ritz, J. F., Braucher, R., Vassallo, R., Prentice, C., Mahan, S., et al. (2011). Slip rate and slip magnitudes of past earthquakes along the Bogd left-lateral strike-slip fault (Mongolia). *Geophysical Journal International*, *186*(3), 897–927. https://doi.org/10.1111/j.1365-246x.2011.05075.x
   Rizza, M., Ritz, J., Prentice, C., Vassallo, R., Braucher, R., Larroque, C., et al. (2015). Earthquake geology of the Bulnay Fault (Mongolia).
- Bulletin of the Seismological Society of America, 105(1), 72–93. https://doi.org/10.1785/0120140119
- Rudnick, R. L., & Fountain, D. M. (1995). Nature and composition of the continental crust: A lower crustal perspective. *Reviews of Geophysics*, 33(3), 267–309. https://doi.org/10.1029/95rg01302
- Ruppen, D., Knaf, A., Bussien, D., Winkler, W., Chimedtseren, A., & von Quadt, A. (2014). Restoring the Silurian to Carboniferous northern active continental margin of the Mongol–Okhotsk Ocean in Mongolia: Hangay–Hentey accretionary wedge and seamount collision. *Gond-wana Research*, 25(4), 1517–1534. https://doi.org/10.1016/j.gr.2013.05.022
- Ruzhentsev, S. V., Badarch, G., & Voznesenskaya, T. A. (1985). Tectonics of the Trans-Altai zone of Mongolia (Gurvansaykhan and Dzolen ranges). *Geotectonics*, 19, 276–284.
- Ruzhentsev, S. V., Pospelov, I., & Badarch, G. (1992). The Inner-Mongolia ophiolitic sutures. *Doklady Akademii Nauk SSSR*, 322(5), 953–958.
- Saraswati, A. T., Cattin, R., Mazzotti, S., & Cadio, C. (2019). New analytical solution and associated software for computing full-tensor gravitational field due to irregularly shaped bodies [Software]. Journal of Geodesy, 93(12), 2481–2497. https://doi.org/10.1007/s00190-019-01309-y
- Schmidt, S., Plonka, C., Götze, H. J., & Lahmeyer, B. (2011). Hybrid modelling of gravity, gravity gradients and magnetic fields. *Geophysical Prospecting*, 59(6), 1046–1051. https://doi.org/10.1111/j.1365-2478.2011.00999.x
- Şengör, A. M. C., Natal'in, B. A., & Burtman, V. S. (1993). Evolution of the Altaid tectonic collage and Paleozoic crustal growth in Eurasia. *Nature*, 364(6435), 299–307. https://doi.org/10.1038/364299a0
- Sheldrick, T. C., Barry, T. L., Van Hinsbergen, D. J. J., & Kempton, P. D. (2018). Constraining lithospheric removal and asthenospheric input to melts in Central Asia: A geochemical study of Triassic to Cretaceous magmatic rocks in the Gobi Altai (Mongolia). *Lithos*, 296–299, 297–315. https://doi.org/10.1016/j.lithos.2017.11.016
- Shen, W., Ritzwoller, M. H., Schulte-Pelkum, V., & Lin, F. C. (2013). Joint inversion of surface wave dispersion and receiver functions: A Bayesian Monte-Carlo approach. *Geophysical Journal International*, 192(2), 807–836. https://doi.org/10.1093/gji/ggs050
- Soejono, I., Buriánek, D., Janoušek, V., Svojtka, M., Čáp, P., Erban, V., & Ganpurev, N. (2017). A reworked Lake Zone margin: Chronological and geochemical constraints from the Ordovician arc-related basement of the Hovd Zone (western Mongolia). *Lithos*, 294–295, 112–132. https://doi.org/10.1016/j.lithos.2017.08.014
- Soejono, I., Čáp, P., Míková, J., Janoušek, V., Buriánek, D., & Schulmann, K. (2018). Early Palaeozoic sedimentary record and provenance of flysch sequences in the Hovd Zone (western Mongolia): Implications for the geodynamic evolution of the Altai accretionary wedge system. *Gondwana Research*, 64, 163–183. https://doi.org/10.1016/j.gr.2018.07.005
- Soejono, I., Janoušek, V., Peřestý, V., Schulmann, K., Svojtka, M., Hanžl, P., et al. (2023). From Rodinian passive margin to peri-Siberian continental arc: Evidence from the multiphase Neoproterozoic–early Paleozoic magmatic record of the Zavkhan Block in the Mongolian Collage. Gondwana Research, 121, 344–367. https://doi.org/10.1016/j.gr.2023.05.001
- Štípská, P., Peřestý, V., Soejono, I., Schulmann, K., Kylander–Clark, R. A., Aguilar, C., et al. (2023). Anticlockwise metamorphic paths at ca. 890–790 Ma from the NE Baidrag block, Mongolia, indicate back-arc compression at the Rodinia periphery. *Geoscience Frontiers*, 14(2), 101520. https://doi.org/10.1016/j.gsf.2022.101520
- Štípská, P., Schulmann, K., Lehmann, J., Corsini, M., Lexa, O., & Tomurhuu, D. (2010). Early Cambrian eclogites in SW Mongolia: Evidence that the Palaeo-Asian Ocean suture extends further east than expected. *Journal of Metamorphic Geology*, 28(9), 915–933. https://doi. org/10.1111/j.1525-1314.2010.00899.x
- Sukhbaatar, T., Lexa, O., Schulmann, K., Aguilar, C., Štípská, P., Wong, J., et al. (2022). Paleozoic geodynamics and architecture of the southern part of the Mongolian Altai Zone. *Tectonics*, 41(8), e2022TC007498. https://doi.org/10.1029/2022tc007498
- Tiberi, C., Deschamps, A., Déverchère, J., Petit, C., Perrot, J., Appriou, D., et al. (2008). Asthenospheric imprints on the lithosphere in Central Mongolia and Southern Siberia from a joint inversion of gravity and seismology (MOBAL experiment). *Geophysical Journal International*, 175(3), 1283–1297. https://doi.org/10.1111/j.1365-246x.2008.03947.x
- Tiberi, C., Ebinger, C., Ballu, V., Stuart, G., & Oluma, B. (2005). Inverse models of gravity data from the Red Sea-Aden-East African rifts triple junction zone. *Geophysical Journal International*, 163(2), 775–787. https://doi.org/10.1111/j.1365-246x.2005.02736.x
- Tomurtogoo, O., Windley, B. F., Kröner, A., Badarch, G., & Liu, D. Y. (2005). Zircon age and occurrence of the Adaatsag ophiolite and Muron shear zone, central Mongolia: Constraints on the evolution of the Mongol-Okhotsk ocean, suture and orogen. *Journal of the Geological Soci*ety, 162(1), 125–134. https://doi.org/10.1144/0016-764903-146
- Tsukada, K., Nakane, Y., Yamamoto, K., Kurihara, T., Otoh, S., Kashiwagi, K., et al. (2013). Geological setting of basaltic rocks in an accretionary complex, Khangai–Khentei Belt, Mongolia. *Island Arc*, 22(2), 227–241. https://doi.org/10.1111/iar.12028
- Valtr, V., & Hanzl, P. (2008). Geophysical cross-section through the Bogd fault system in the area of the Chandman rupture, SW Mongolia. *Journal of Geosciences*, 53(2), 193–200. https://doi.org/10.3190/jgeosci.023
- Walker, R. T., Nissen, E., Molor, E., & Bayasgalan, A. (2007). Reinterpretation of the active faulting in central Mongolia. *Geology*, 35(8), 759–762. https://doi.org/10.1130/g23716a.1
- Wang, T., Guo, L., Zhang, J., Tong, Y., Wang, Y., Hao, L., et al. (2019). Map of granitoids and related rocks of Asia. Retrieved from http:// igcp662.org.cn/maps?article\_id=63
- Wilhem, C., Windley, B. F., & Stampfli, G. M. (2012). The Altaids of Central Asia: A tectonic and evolutionary innovative review. Earth-Science Reviews, 113(3–4), 303–341. https://doi.org/10.1016/j.earscirev.2012.04.001
- Xiao, W., Windley, B. F., Han, C., Liu, W., Wan, B., Zhang, J., et al. (2018). Late Paleozoic to early Triassic multiple roll-back and oroclinal bending of the Mongolia collage in Central Asia. *Earth-Science Reviews*, 186, 94–128. https://doi.org/10.1016/j.earscirev.2017.09.020

- Xiao, W., Windley, B. F., Sun, S., Li, J., Huang, B., Han, C., et al. (2015). A tale of amalgamation of three permo-triassic collage systems in central Asia: Oroclines, sutures, and terminal accretion. *Annual Review of Earth and Planetary Sciences*, *43*, 477–507.
- Xiao, W. J., Windley, B. F., Yuan, C., Sun, M., Han, C. M., Lin, S. F., et al. (2009). Paleozoic multiple subduction-accretion processes of the southern Altaids. American Journal of Science, 309(3), 221–270. https://doi.org/10.2475/03.2009.02
- Yakubchuk, A. S., Shatov, V. V., Kirwin, D., Edwards, A., Tomurtogoo, O., Badarch, G., & Buryak, V. A. (2005). Gold and base metal metallogeny of the Central Asian orogenic supercollage. *Economic Geology*, 100th AnniversaryVolume, 1069–1096.
- Yarmolyuk, V. V., Kovach, V. P., Kozakov, I. K., Kozlovsky, A. M., Kotov, A. B., & Rytsk, E. Y. (2012). Mechanisms of continental crust formation in the Central Asian Foldbelt. *Geotectonics*, 46(4), 251–272. https://doi.org/10.1134/s001685211204005x
- Yarmolyuk, V. V., Kovalenko, V. I., Kozakov, I. K., Sal'nikova, E. B., Bibikova, E. V., Kovach, V. P., et al. (2008). The age of the Khangai batholith and the problem of batholith formation in Central Asia. *Doklady Earth Sciences*, 423(1), 1223–1228. https://doi.org/10.1134/ s1028334x08080096
- Yarmolyuk, V. V., Kozlovsky, A. M., Savatenkov, V. M., Kovach, V. P., Kozakov, I. K., Kotov, A. B., et al. (2016). Composition, sources, and geodynamic nature of giant batholiths in Central Asia: Evidence from the geochemistry and Nd isotopic characteristics of granitoids in the Khangai zonal magmatic area. *Petrology*, 24(5), 433–461. https://doi.org/10.1134/s0869591116050064
- Yarmolyuk, V. V., & Kuzmin, M. I. (2012). Late Paleozoic and Early Mesozoic rare-metal magmatism of Central Asia: Stages, provinces, and formation settings. *Geology of Ore Deposits*, 54(5), 313–333. https://doi.org/10.1134/s1075701512050054
- Zhao, H., Wang, P., & Huang, Z. (2021). Lithospheric structures beneath the western Mongolian Plateau: Insight from S wave receiver function. Journal of Asian Earth Sciences, 212, 104733. https://doi.org/10.1016/j.jseaes.2021.104733
- Zhu, L., & Kanamori, H. (2000). Moho depth variation in southern California from teleseismic receiver functions. Journal of Geophysical Research: Solid Earth, 105(B2), 2969–2980. https://doi.org/10.1029/1999jb900322
- Zonenshain, L. P. (1973). The evolution of Central Asiatic geosynclines through sea-floor spreading. *Tectonophysics*, 19(3), 213–232. https://doi.org/10.1016/0040-1951(73)90020-6
- Zonenshain, L. P., Kuzmin, M. I., & Natapov, L. M. (1990). Geology of the USSR: A plate tectonic synthesis (p. 334).
- Zonenshain, L. P., & Kuzmin, M. I. (1978). Khantaishir ophiolite complex in western Mongolia and ophiolite problem. *Geotectonics*, 1, 19–42.
  Zorin, Y. A. (1999). Geodynamics of the western part of the Mongolia-Okhotsk collisional belt, Trans-Baikal region (Russia) and Mongolia. *Tectonophysics*, 306(1), 33–56. https://doi.org/10.1016/s0040-1951(99)00042-6
- Zorin, Y. A., Belichenko, V., Turutanov, E., Kozhevnikov, V., Ruzhentsev, S., Dergunov, A., et al. (1993). The South Siberia-Central Mongolia transect. *Tectonophysics*, 225(4), 361–378. https://doi.org/10.1016/0040-1951(93)90305-4
- Zorin, Y. A., Mordvinova, V. V., Turutanov, E. K., Belichenko, B. G., Artemyev, A. A., Kosarev, G. L., & Gao, S. S. (2002). Low seismic velocity layers in the Earth's crust beneath Eastern Siberia (Russia) and Central Mongolia: Receiver function data and their possible geological implication. *Tectonophysics*, 359(3), 307–327. https://doi.org/10.1016/s0040-1951(02)00531-0